# **MASTER OF PROFESSIONAL PRACTICE**

# AN INVESTIGATION INTO HOW AND WHY INFORMATION SHOULD BE STORED AND SHARED FOR PEOPLE WITH HIGH AND COMPLEX SUPPORT NEEDS.

## **BRIDGET MEYER**



## **Dedication:**

To Riley Bell and all the Riley Bell's in our world.



Figure 1: Archery and riding on the back of Grandad's ute during school camp. Front page. Riley listening to instructions.

## Acknowledgements:

I wish to thank the following people who have supported me during my Master of Professional Practice journey.

To Riley Bell. Riley you may not know this but our time together on your school camp has been a huge catalyst in getting this research project both underway and completed. When I felt overwhelmed at what lay ahead in bringing this piece of work together, I thought about you. I pictured you in the trail rider when we got to the fork in the track and all your classmates went one way and your mum, grandad, grandma, teacher aide and I went the other. If only the teachers in charge had known just how capable you were. If only they had known what you had done the previous year. If only they had asked some questions. A lot happened that day for me Riley, and I will be forever grateful to you. Thankyou Riley.

To Rowena, Gerald and Riley's Nana and Grandad. Thank you for all you do.

To all the research participants. Thank you for your time and honesty in sharing what for some of you were very personal thoughts and challenges. I hope through this research we can find a way to develop a tool that will allow information to be stored and shared across your networks making life more meaningful for all.

To Graeme Burborough from The Tarn Group for making the time to create the demo site that gave me a starting point to share with the research participants. I'm excited to share the research findings with you Graeme.

To Sue Gates and Cherie McConville who created a safe listening space as I began working with the building blocks of what was to become my thesis. Being able to literally pour my thoughts out on why I thought this research project needed to happen and you both asking some great triggering questions was extremely helpful. Thank you both.

To my colleagues at the Halberg Foundation for picking up my work while I took leave to complete my master's thesis. To Shelley McMeeken and Lucinda Searchfield for your understanding and support as I undertook this journey. I am truly appreciative of being able to step away from the daily happenings within the world of Halberg and give my attention to my master's thesis. Thank you all.

To Dr Jo Kirkwood for your encouragement to keep going and immediate responses to the never-ending questions I had. You never left me waiting on a response which I appreciated enormously. The task you have of making sense of your learner's work-based learning projects cannot be easy however you were able to unravel the chaos of my thinking and help place those thoughts into a format that brings a practioner's world and a researcher's world together. Thank you, Jo.

To Dr Glenys Forsyth for ensuring standards were kept high and that my dots were where they ought to be. Your eye for detail, although an annoyance at the times (I mean that in a loving way), has ensured academic integrity has been maintained. Every time I see a fine tooth comb from now on I will think of you. Thank you, Glenys.

To CCS -Disability Action particularly Melissa Smith for understanding the importance of this research project and being my surrogate workplace supporter. I know the numbers of our most high and complex kids are relatively small but you understood that they too need a voice and embraced and nurtured this project as your own when I first presented it to you. Thank you, Melissa.

Finally, to my partner Professor Rhonda Rosengren. Thank you for creating the space for me to complete my Master of Professional Practice. I truly appreciate having you on my side. You didn't once question why the fire had gone out or if we would be eating dinner together on any given night knowing that I was in 'the zone'. You more than helped bring clarity to what has been a muddily and at times trying journey. Now to get some of our life back. I'm unsure if I can promise you that I am done with this academic learning gig but that is a conversation we can have offline. Thank you, Rhonda.

STUDENT NAME:	Bridget Tracey Meyer
STUDENT ENROLEMENT NUMBER:	2006174

ATTESTATION OF AUTHORSHIP:

"I hereby declare that this submission is my own work and that, to the best of my knowledge and belief it contains no material previously published or written by another person (except where explicitly defined in the acknowledgments), nor material which to a substantial extent has been submitted for the award of any other degree or diploma of an institution of higher learning".

NAME:

Bridget Meyer

SIGNATURE:

## Table of Contents:

N	ASTER	OF PROFESSIONAL PRACTICE	
D	edicatio	on:	2
A	cknowle	edgements:	3
Τa	able of	Contents:	6
Fi	gures:		9
Та	ables:		
G	ossary		
1	CHA	PTER ONE: Introduction	
	1.1	Executive Summary	
	1.2	Introduction to the researcher	
	1.3	Introduction to my Thesis	
	1.4	Background to research	
	1.5	A day in the life of	
	1.6	Learning Outcomes	20
	1.7	Reflective Practitioner	Error! Bookmark not defined.
	1.8	Reflective Commentary - Introduction	Error! Bookmark not defined.
	1.9	Insight One: Early Education	Error! Bookmark not defined.
	1.10	Insight Two: Education now	Error! Bookmark not defined.
2	СНА	PTER TWO: Literature Review	21
	2.1	Review of Literature - Introduction	21
	2.2	United Nations Convention and National Legislation	
	2.3	Lived Experiences – Advocacy and Empowerment	
	2.4	Medical Model of Disability	
	2.5	Social Model of Disability	
	2.6	Collaboration	
	2.7	Technology in an unspoken world	
	2.8	Storing and Sharing Information	
	2.9	Summary of Literature Review	
	2.10	Insight Three: Dusting myself off	Error! Bookmark not defined.
_			
3	CHA	PTER THREE: Methodology and Method	

	3.2	Qualitative Vs Quantitative	
	3.3	Research Method	
	3.4	Concept screen shots of website	49
	3.5	Research Participants	51
	3.6	Research Questions	Error! Bookmark not defined.
	3.7	Insight Four: Graphs, Tables and writing by hand	53
	3.8	Insight Six: Uncharted waters.	Error! Bookmark not defined.
4	CHA	PTER FOUR:	54
	4.1	Introduction to Findings:	54
	4.2	Reaction	55
	4.3	Discussion - Reaction	57
	4.4	Current Situation	
	4.5	Discussion – Current Situation:	
	4.6	Advocacy	65
	4.7	Discussion - Advocacy	67
	4.8	Empowerment	
	4.9	Discussion - Empowerment	74
	4.10	Collaboration	
	4.11	Discussion - Collaboration	77
	4.12	Resources	
	4.13	Discussion - Resources	
	4.14	Accessibility	
	4.15	Discussion - Accessibility	
	4.16	Security	
	4.17	Discussion - Security	
	4.18	Barriers	
	4.19	Discussion - Barriers	
	4.20	Summary of Findings	
	4.21	INSIGHT SIX: Learning and other muses:	
5	CHA	PTER FIVE:	91
	5.1	Evaluation of Study	91
	5.2	Recommendations	92
	5.3	What is next for me?	Frror! Bookmark not defined.

	5.4	Insight Seven: Throwing a grenade	Error! Bookmark not defined.
	5.5	Insight Eight: Where to from here?	Error! Bookmark not defined.
6	REF	ERENCES:	93
7	APP	ENDICES:	
	7.1	Information for Research Participants	
	7.2	Ethics template to monitor changes	
	7.3	Letter of support from CCS - Disability Action:	
	7.4	Keeping track of required information	
	7.5	Sample of steps taken to analyse transcripts	
	7.6	Example of journal entries	Error! Bookmark not defined.
	7.7	Curriculum Vitae	
	7.8	Review of Learning – Course one	
	7.9	Learning Agreement – Course two	



Figure 2: Poppy having the best time on school camp.

## Figures:

FIGURE 1: ARCHERY AND RIDING ON THE BACK OF GRANDAD'S UTE DURING SCHOOL CAMP. FRONT PAGE. RILEY	
LISTENING TO INSTRUCTIONS	
FIGURE 2: HAVING THE BEST TIME EVER ON SCHOOL CAMP	ì
FIGURE 3: INITIAL CONCEPT OF IDEA	
FIGURE 4: GOAL IMAGE	I
FIGURE 5: DR BRENE BROWN - BRAVING THE WILDERNESS WORKSHEETBRROR! BOOKMARK NOT DEFINED.	,
FIGURE 6: AUTHOR'S OWN DESCRIPTION OF DEVELOPMENT OF LEARNING AND SHIFT IN THINKING! ERROR!	
BOOKMARK NOT DEFINED.	
FIGURE 7: MY OWN - ORGANISATIONS AND LEGISLATION IN SUPPORT OF PEOPLE WITH DISABILITIES26	
FIGURE 8: MEDICAL AND SOCIAL MODEL DIAGRAM	l
FIGURE 9: MY OWN FIGURE WITH ADDITIONAL IDEAS ON THE BENEFITS OF COMMUNITY CONVERSATIONS	
FIGURE 10; ASSISTIVE TECHNOLOGY; IMAGE FROM ASSISTIVE TECHNOLOGY FACEBOOK PAGE35	
FIGURE 11: THE ACTION RESEARCH CYCLE41	
FIGURE 12: IMAGE OF QUANTITATIVE VS QUALITATIVE DATA43	
FIGURE 13: FLYER RECRUITING INTERVIEW PARTICIPANTS	
FIGURE 14: LANDING PAGE	
FIGURE 16: ADMINISTRATORS PAGE	ł
FIGURE 17: OPTIONS OF SIX PLACES TO HOUSE INFORMATION	
FIGURE 18: WINDOW TO HOUSE INFORMATION	
FIGURE 19: LIST OF PEOPLE WITHIN PERSON'S CIRCLE OF SUPPORT50	ł
FIGURE 20: DEMONSTRATION ON HOW TO DRAG AND DROP INTO APPROPRIATE FILE	
FIGURE 21: WIZARD TO SAY INFORMATION HAS BEEN SHARED	
FIGURE 22: IMAGE FOR RESEARCH QUESTIONS	
FIGURE 23: SCREENSHOT OF HOW I DO THINGS USING VISUAL CLUES ERROR! BOOKMARK NOT DEFINED.	,
FIGURE 24: SCREENSHOT OF WRITING NOTES FOR LITERATURE REVIEW	
FIGURE 26: MIND MAP OF POTENTIAL JOURNEY OF MPP PROCESS	
FIGURE 27: ACTION RESEARCH MODEL	
FIGURE 28: EXAMPLE OF PARALLEL PROCESS	

## Tables:

TABLE 1: MY OWN ADAPTED TABLE OF TIMELINE OF SIGNIFICANT EVENTS WITHIN NEW ZEALAND EDUCATION	
SYSTEM	25
TABLE 2: ADAPTED FROM JUSTIN A. HAEGELE - COMPARISON BETWEEN THE MEDICAL AND SOCIAL MODEL	
DISCOURSE	31
TABLE 3: EXAMPLE OF SPREADSHEET SHOWING HOW INFORMATION WAS CAPTURED.	48
TABLE 4: ETHICS APPLICATION INFORMATION	101
TABLE 5: ETHICS APPLICATION INFORMATION	102
TABLE 6: ETHICS APPLICATION INFORMATION.	
TABLE 7: ETHICS APPLICATION INFORMATION	104
TABLE 8: ETHICS REVISION PROOF OF CHANGES	105
TABLE 9: TABLE TO KEEP TRACK OF INFORMATION REQUIRED OF JOURNAL ENTRIES	108
TABLE 10: EXAMPLE OF STEPS TAKEN TO ANALYSE TRANSCRIPTS	110

## Glossary:

Master of Professional Practice
Bachelor of Social Services
United Nations Convention of Rights of Persons with Disability
Disability Information Service
United Nations
Independent Monitoring Mechanism
Intellectually Handicapped
Education Review Office
Union of the physically impaired against segregation
World Health Organisation
Interprofessional Collaboration
Interprofessional Education
Interprofessional Community Practice
Assistive Technology
Chief Executive
High and Complex Needs

Note: Permission has been received to use all photos within my thesis.

### **1 CHAPTER ONE: Introduction**

#### 1.1 Executive Summary

My interest in completing my thesis through the Master of Professional Practice (MProfPrac) programme was inspired during my time studying within the Capable NZ undergraduate programme which is offered through the Otago Polytechnic, New Zealand. The Capable programme encourages learners, like myself, to continue to remain within their workplaces whilst learning to become critical thinkers and undertaking a research project that relates directly to their respective disciplines. This was something I was extremely interested in pursuing and although I was not supported directly through my workplace, the Halberg Foundation, and have carried out this project in my own time and at my own cost, I was able to gain support within the disability sector to ensure proper process and integrity was maintained.

The purpose of my thesis was to investigate the concept of a storing and sharing website for people with high and complex support needs within the disability sector. I was looking to gain clarity through this process and an understanding on the mechanisms that are currently available for individuals and the benefits of such. Through this process I was also seeking out the barriers that such a concept may present.

The opportunity to place this work-based concept of a storing and sharing website into an academic environment has allowed me to cast a unique lens over it. The rigour and robustness in acquiring knowledge through this method has been both enlightening and at times confronting. The findings of my research place the concept in a good position to take forward however I do recommend the intricate design and logistics of accessible features need to be explored further with a recommendation to study specific areas of the findings in more depth.

> "Education is the most powerful weapon which you can use to change the world" Nelson Mandela

#### 1.2 Introduction to the researcher

I am currently employed as a Lead Adviser for the Halberg Disability Sport Foundation, a role I have been doing for the past 13 years. My core role for the Foundation is focused on breaking down barriers and creating opportunities to encourage people of all levels of ability to participate in sport and recreation in a meaningful way alongside their peers. The strengths I bring to this role are in being innovative as well as being persistent around ways in which we can turn what some people perceive as an obstacle into what I would describe as an opportunity.

Through my role I have had the opportunity to develop several programmes that have been implemented both regionally and nationally. The Adaptive Sport Mentoring programme was presented in Paris in 2011 at the International Symposium of Adapted Physical Activity whereby interest from Belgium and America was established. The programme was based on mentoring young people with physical challenges into an inclusive sport or recreational environment. The Adaptive Sport Mentoring programme has been successfully delivered through the University of Otago since 2008. A second example of innovation is I am currently working on a website that allows individuals to make informed decisions on outdoor places and spaces depending on their level of ability. I am leading this project on behalf of the Foundation and now have the Department of Conservation, Be. Accessible and Fabriko as project partners. I am particularly interested in creating programmes and platforms that encourage both independent decision making and independent engagement.

Regarding my personal values and philosophy these have developed over the course of my lifetime and through different periods of my life have been adjusted. Our family, friends and community and the experiences we have had, all contribute to my sense of who I am and how I view the world. In the social service sector, I am often working with people who are vulnerable and/or may have lifestyles that are different than my own. It is important to those of us working in this sector that we are aware of our own personal values and are mindful not to impose them on the people we are working alongside. My values are something I hold in high regard. My values guide the way I live my life and the decisions that I make along the way both on a personal and professional level.

12

- 1) Being there for my family.
- 2) Working to promote equality and justice in our society.
- 2) Being emotionally happy.
- 4) Making a positive contribution to the community.
- 5) Continuing to learn new things.
- 6) Being accountable and responsible.
- 7) Being creative and thinking beyond traditional ideas to bring about change.
- 8) Being honest and truthful.
- 9) Manners for example, being on time.



My values act as a touchstone that I can come back to regularly. They act as my compass that keeps me operating between North and True North. I don't believe I have a compass for work and for home life, I believe they are one and the same and I conduct myself with these same values regardless of who I am engaging with.

#### 1.3 Introduction to my Thesis

This introduction gives a brief overview of the structure of my thesis along with information that is discussed within each chapter. My thesis has five chapters (1) Introduction, (2) Review of Literature, (3) Methodology and Method, (4) Research Findings, (5) Evaluation of Study and Recommendations. At the end of each chapter, I have shared my reflective commentary of learnings gained through the MProfPrac process. Each commentary gives an insight into the various stages of my learning journey as well as other occurrences that were adding colour to the MProfPrac process. Within chapter one I will also introduce *a day in the life* of a family with two young men who have high and complex support requirements.

Chapter One includes an executive summary explaining the purpose of my research followed by an introduction of myself as the researcher where I give an insight into my values and what is important to me both professionally and personally. The background to my research is introduced along with the learning outcomes I am looking to achieve through this process. Within chapter One, I have presented my first two insights. My first insight is my early education where I have touched on some of the challenges I had during my schooling years. The second insight discusses what my education looks like now.

Chapter Two is my review of literature. Within this chapter I have looked at literature focusing on several areas that I believed relevant to my research question. These areas are (1) United Nations Convention and National Legislation (2) Lived Experiences – Advocacy and Empowerment (3) Medical Model of Disability (4) Social Model of Disability (5) Collaboration (6) Technology in an unspoken world (7) Storing and sharing information. I have given an overview of what each of these areas will encompass at the beginning of the literature review as well as a summary at the conclusion of the literature review. Chapter Two also includes my third insight titled 'dusting myself off'.

Chapter Three outlines both the methodology I used for this research and the methods used in my research. I also share the screenshots of the website concept that were used during the interview process and other pieces of relevant information. The research participants are introduced in Chapter Three as well as the research questions. I have included insights four and five into Chapter Three.

Chapter Four discusses the research findings as well as the discussion on each point of interest. The findings have been discussed through the themes that emerged during the interview process. I have also included insight six into chapter four.

Chapter Five brings my thesis to a conclusion with the evaluation of the study being discussed along with my future recommendations. Insight seven and eight are included in Chapter Five.

14

#### 1.4 Background to research

The landscape in which individuals with disabilities need to traverse in terms of funding and resources are complex regardless of where on the spectrum an individual falls. For individuals that acquire an injury as opposed to a congenital disability i.e born with that disability, the framework in which they operate is very different. Many of the young people I work alongside are supported through an intersectoral partnership between the Ministry of Health, Ministry of Education and Oranga Tamariki. Each of these organisations engages with and invests financially into a strategy called High and Complex Needs (HCN), *Me mahi tahi tatou*. The goals of the HCN strategy include collaboration across joint systems, encouraging intersectoral practice and the improvement of individualised service packages. This partnership and the funding that is associated with it is supporting individuals with the highest and most complex requirements. The number of individuals that receive this level of tailored service according to the 2005 Intersectoral Strategy for Children and Young People with High and Complex Needs, *Te Kahu Tauahi Whānau*, document is around 150. (Foreword 2 Introduction 3, n.d.). The strategy hosts several different outcomes they are looking to achieve of which four are particularly pertinent to this research.

- 1) Strong educational foundation for lifelong learning.
- Social interaction. Established relationships with peers. Engages positively in social situations and participates in the community.
- Safety and security. The family/whānau/caregivers are supportive, well-informed and meet the needs of the child.
- 4) HCN outcomes for the sector are directed towards relationship building and setting up systems so that service providers are better able to work together.

The other avenue whereby support is accessed is through the Accident Compensation Corporation (ACC). The numbers of individuals that fit the high and complex criteria, particularly children, is not high. To be funded through the ACC model you need to have acquired your disability through an accident or medical misadventure. The support, both financially and resource-wise, is more readily available if an individual does fit within this funding stream. For the purposes of my thesis, I have attempted to research accurate data sets that represent individuals with high and complex needs; however, these are not currently available through Statistics New Zealand. The most up-to-date sets claims 24 percent of the population has a disability; however, this includes a large proportion of people over 65 years who wear either glasses or hearing aids. According to Statistics NZ, 11 per cent of children under 15 years had a disability of which a learning disability was the most common (Statistics New Zealand., n.d.).

Through my role as a lead adviser for the Halberg Disability Sport Foundation, I have been in a position where I have observed the challenges that are encountered by schools, community support organisations and sports clubs, where potential inclusive opportunities are presented yet not embraced. An example of this is when support is being provided to an individual and solutions will be created on how to modify a situation or piece of equipment. Success in that moment has been conceived and a positive outcome obtained. The 'problem' is that no record is made of that moment. This loss of a valuable moment in record keeping is seen as an ongoing problem. With the absence of significant information, the support staff are unable to plan, modify or adapt in any meaningful engagement for the student they are supporting to become involved with the class. The sharing of this information, had it been stored and then shared, could potentially bring an inclusive opportunity for a student. To gain an understanding of the number of people that work alongside individuals see figure 3.

My interest in attempting to find a solution comes from conversations I have had with both individuals with additional support needs, their families, schools and community support organisations, on how information can be stored and shared as a child transitions through their school years and post school. For individuals who are unable to advocate for themselves this information can be vital as they work towards creating a meaningful life experience.

In my Review of Learning document, I refer to several important goals that I would like to achieve through the MProfPrac process. Gaining clarity on the inclusive education space within New Zealand, finding a platform in which I can align my own philosophy of inclusive practice, working towards a solution focused approach that encourages collaboration and looking closely at my own professional practice.

In having a good understanding of the High and Complex Needs systems that are currently in place; the desire of organisations to work more collaboratively, and the challenges that are presented on a sometimes-daily basis for the individuals and families, I work alongside I thought there must be a way we can do things differently. If there was a way that information could be shared readily, with permission, housed safely to build a rich vault of ideas to draw on, then noone is going to be worse off. The idea of Facebook, Google docs, Twitter all in one place could possibly, if designed well, be a valuable resource.



Figure 3: initial concept of idea

#### 1.5 A day in the life of....

In order to bring as much context to my thesis as possible, I have sought permission to share what a typical day looks like for a family whom have two young men with high and complex support needs. The ages of Harry and Tom are 21 years and 18 years respectively. Harry has recently left school, while Tom is in year 13. For individuals with high and complex support needs, they can stay at school until the age of 21.

The purpose in sharing this insight is to begin to build a picture of what it is like when an individual is unable to advocate for him or herself yet has numerous support people interacting with them throughout the day. The day begins first thing in the morning, when the early morning caregiver arrives; extends through the day, when the teacher aides, gym instructors, physios, afternoon support worker have input; and moves into the evening, when the support worker leaves the family home around 9.00pm. This example is over one day. However, over the course of the week, there are different support workers coming and going along with different people who support the various activities that take place. For Tom, who is still at school, the teacher aides are relatively consistent; however, due to his additional support needs, i.e sign language, it is very challenging when they are away.

This information is used with permission.

6.30 am	Mum gets up and makes breakfast and lunches.
7.00am	The morning caregiver arrives, and she and Mum get both boys, toileted, dressed, hoisted/transferred into wheelchairs.
7.30am	Being fed breakfast.
8.00am	Loaded into van ready to drive to school.
8.30am	Drop Tom off at school. Return home and toilet Harry before his support worker arrives.
9.30am	Together they head out to catch bus to the gym.
10.45am	Mum goes to physio appointment for workplace injury through lifting.
12.00pm	Harry and support worker meet Mum at Dental School. Support worker finishes and Harry attends a dental apt.
1.00pm	Harry and Mum head home, mum toilets Harry and then walks the dog.
2.30pm	Harry and Mum head to school to pick up Tom

3.00pm	Load Tom in van and head home via supermarket.		
3.30pm	Get home, unload, and have afternoon tea.		
3.45pm	Physiotherapist and wheelchair rep arrives to look at Harry in his chair. Afternoon caregiver arrives and changes Tom out of his uniform and gets him on toilet.		
4.30pm	Boys hoisted into their armchairs.		
4.45pm	Physios and caregivers leave.		
	TV watching 😊		
6.15pm	Dad arrives home and we have dinner.		
7.00pm	We all relax and watch an Alaskan Coast Guard show on Netflix.		
8.00pm	The night-time caregiver arrives. Toileting and bedtime routine.		
9.00pm	Caregiver leaves.		

#### 1.6 Learning Outcomes

The goal of my research is to investigate the concept of a storing and sharing website for people with high and complex support needs. This website concept would encourage ideas and information to be shared across a person's support networks. The website would be a platform to house relevant information that would build over time creating a dossier of rich information. I will investigate whether the concept of such a website is something that could be used by the sector and to initiate conversations with the research participant group on how such an idea of incorporating technology and encouraging interprofessional collaborative practice can benefit the disability sector as well an gaining an insight into the challenges that such a concept may present.

Secondly, my MProfPrac journey has allowed me to gain a more in-depth understanding of my own professional practice and given me time to consider where to from here. The reflective component of the MProfPrac course has enabled me the opportunity to recalibrate my own practice and question the direction I am taking within the social service sector.



Figure 4: Goal image.

("What I've Learned about Goals and Objectives - Listen Technologies," n.d.)

### **2** CHAPTER TWO: Literature Review

### 2.1 Review of Literature - Introduction

In this chapter, I summarise the areas of advocating for individuals with high and complex needs, lived experiences and empowerment, understanding interprofessional collaboration, using technology in an unspoken world and information storing and sharing websites. These areas of literature will guide my research question. I will also outline the International legislation from the United Nations Convention of the Rights of Persons with Disability along with National legislation that impacts young people with high and complex support needs within the education curriculum. I am seeking literature from these fields to be informed on how advocacy is fostered for individuals with high and complex needs; how individuals are empowered through their lived experiences; how collaboration works across varies sectors; how assistive technology can be of benefit or of detriment to an individual; and what needs to be considered in storing and sharing information.

The structure of this chapter will begin with an overview of the United Nations Convention on the Rights of Persons with Disabilities with a focus on Article 21, Article 24 and Article 30. I will summarise the history of inclusive education in New Zealand and introduce legislation from the New Zealand Education Act (1889). I will draw on literature that looks at how this legislation has impacted children with disabilities. In the second section, I begin by drawing on literature that gives a voice to the lived experiences of individuals through advocacy. Section two will also look to the literature to understand empowerment for individuals with high and complex needs and how this can inform and guide our practice. I will also introduce the medical model of practice and the social model of practice. After outlining the literature on lived experiences and introducing the medical and social model of practice, I will draw on literature that explains varies forms for collaboration in both a medical, education and community context. In the final section, I describe a form assistive technology that can be used in an unspoken world and examine the literature that discusses both the positive and negative outcomes of engaging with assistive technology. I introduce and draw on the literature of two recently-developed websites and discuss the findings of one of the sites. I will close this chapter by revisiting the main points that have been identified in the review.

#### 2.2 United Nations Convention and National Legislation

The United Nations was founded in 1945 with the aim of achieving both peace and security at an international level as well as promoting social progress and engaging in human rights. Of the 195 countries in the world, 193 countries are members of the United Nations (History of the UN | United Nations Seventieth Anniversary, n.d.). There are four main purposes or focus areas of the United Nations and the work it carries out; (1) to develop peace throughout the world; (2) to develop friendly relations among nations; (3) to be a centre for harmonizing the actions of nations to achieve these goals and, lastly, (4) to help nations work together to improve the lives of poor people, to conquer hunger, disease and literacy and to encourage respect for each other's rights and freedoms. This final point, "to encourage respect for each other's rights and freedoms. This research (History of the UN | United Nations Seventieth Anniversary, n.d.).

At an international level, the Convention on the Rights of Persons with Disabilities sits within the Division for Social Inclusive Social Development which, in turn, comes under the United Department of Economic and Social Affairs. The Convention on the Rights of Persons with Disabilities has 50 Articles relating to various aspects and fundamental rights of a person with a disability ("Convention on the Rights of Persons with Disabilities - Articles | United Nations Enable," n.d.). The global purpose of the Convention is to "promote, protect and ensure the full and equal enjoyment of all human rights and fundamental freedoms by all persons with disabilities, and to promote respect for their inherent dignity" ("Article 1 - Purpose | United Nations Enable," n.d.). There are three articles that are of relevance to this research. Article 21 – Freedom of expression and opinion, and access to information. Article 21 has five key points that highlight various ways in which information can be accessed by people with disabilities. Points B and C are of significance and relevant in that both alternative forms of communication

as well as having access to information and services are recognised as a right ("Article 21 -Freedom of expression and opinion, and access to information | United Nations Enable," n.d.). The importance of education is detailed within article 24. The overarching statement of article 24 identifies an individual's right to an education without discrimination and delivered in such a way that equal opportunities are presented for all learners. This article is divided into five areas, with each area giving guidance to an individual's right within the education system and what can be expected in terms of support from both a human resource and infrastructure viewpoint. For the purposes of this research, however, point 2.e, which identifies the important of individualised support, "consistent with the goal of full inclusion;" and 3.c, stating that education is delivered in the most "appropriate mode and means of communication," are the most noteworthy ("Article 24 - Education | United Nations Enable," n.d.).

The final article within the United Nations Convention on the Rights of Persons with Disabilities is Article 30 – Participation in cultural life, recreation, leisure and sport. Article 30 pertains to accessing a variety of opportunities within an individual's community. Article 30 is divided in five sections with various points under each. The common wording throughout each of these sections is the statement of "on an equal basis with others;" therefore, implying that the accessing of cultural experiences, recreation, leisure and sport will be delivered "on an equal basis with others". Article 30 hosts several important points in relation the rights of an individual. Point five in Article 30 has the overarching statement of "enabling persons with disabilities to participate on an equal basis with others in recreational, leisure and sporting activities" ("Article 30 - Participation in cultural life, recreation, leisure and sport | United Nations Enable," n.d.) With the best of intentions however the ongoing implementation to enforce these articles and the rights within them is a constant challenge (Lyons, 2013, Selvaraj, 2015).

The Office for Disability Issues oversees the Convention on the Rights of Persons with Disabilities by providing a platform that gives visibility and legitimacy to people living with a disability within Aotearoa. To ensure the legitimacy of the convention, in 2010 the New Zealand Government implemented the Independent Monitoring Mechanism (IMM) and tasked it with evaluating the implementation of the Disability Convention in New Zealand. An example of the challenges and areas of concern in fulfilling the convention have been identified by the IMM as recently as March 2018 with six key areas being targeted as ongoing concerns. Education, access to information and communication are two of the six areas the Independent Monitoring Mechanism have recommended be closely looked at. These two areas are of interest to this research. The New Zealand Government and the Office for Disability Issues is currently guided and working towards the eight goals stated in the 2016 – 2026 New Zealand Disability Strategy ("New Zealand Disability Strategy - Office for Disability Issues," n.d.)

Within the New Zealand education system, an individual with additional support needs is also recognised through specific legislation (Ministry of Education, 2012). The 1989 Education Act was a turning point for young people with disabilities. This legislation gave children the right to enroll in their local school with the expectation that they would receive an education alongside their peers. The act stated that "people who have special educational needs (whether because of disability or otherwise) have the same rights to enroll and receive education at State schools as people who do not – Education Act 1989 s(8)(1)" (Ministry of Education, 2012).

The inclusive education movement was initially implemented by the Labour Government during the 1930's (Selvaraj, 2015). The goal was to create an education system that would cater for all levels of abilities within a state school system. A lofty goal and one which continues to present ongoing challenge. According to Selvaraj, the "historical developments of what became known as Special Education" (p.55) created the foundations for an education system that is deficient in clarity in what "inclusion and inclusive education" really means (p.55). There have been numerous manifestations of what best practice could potentially look like for a child with additional support needs. However, as stated by Selvaraj, "the road to inclusion has not been easy" (p. 57). The rhetoric of ideological inclusive practice versus the reality of bringing inclusion to life as opposed to lip service is also discussed by Lyon (2013). Lyon reaffirms that "the concept of inclusion itself has been largely diluted" (p.239). Although legislation and policies have been put in place "and they have been given the language to speak about inclusion, making inclusion a reality appears problematic and confusing" (p.239). The seemingly rudderless direction and confusion arising from the lack of direction continues to perplex to this day. Many professionals across several sectors including education work hard to grapple with this elusive concept of inclusion as both Selvaraj and Lyons have argued. The ideology of what is being proposed

through legislation and policy and the reality of achieving the utopia of inclusion continues to evade us. To gain an understanding of the timeline that the New Zealand Education system has experienced, Table 1 demonstrates the various actions that have been undertaken over the past 142 years.

ACTION	YEAR	OUTCOME
Centrally Funded System.	1877	Centrally Funded and available to all children.
New Zealand Education Act.	1889	Children with Disabilities the right to enroll at local school
UNCROC – United Nations Conventions on the Rights of the Child.	1993	"You have the right to special education and care if you have a disability as well as the right in this convention, so you can live a full life"
Special Education Policy Guideline.	1995	Children may receive special education if they require additional support.
Special Education 2000.	1996	Aimed to achieve a world class inclusive education for New Zealand over the following decade.
New Zealand Disability Strategy.	2001	The aim of the strategy is to ensure individuals have the rights to inclusion, participation and opportunities.
ERO evaluation of inclusive schools.	2010	Education Review Office evaluated the inclusion of students with high needs in mainstream schools. 20% were fully inclusive and 35% were partially inclusive.
IHC Code for New Zealand Schools.	2010	The IHC codes draws from global and national legislation and policy.
Review of Special Education	2010	2000 responses in relation to the Ministry of Education discussion document.
Success For All – Every School, Every Child.	2010	The vision for all schools to be inclusive by 2014.
Collaboration for Success Individual Education Plans.	2011	New approach to IEPs sees the child as central with potential to learn within class and a whole school setting. Collaborative approach.
Special Education Business Plan.	2011	Eight focus areas including the 'Success For All – Every School, Every Child' initiative.
Ministry of Education Report	2014	Schools engaging in Positive Behaviour for Learning programmes (PB4L).
Ministry of Education Report	2018	Schools continue to engage in PB4L programmes.

("History Of Inclusive Education In New Zealand Timeline | adapted from Preceden," n.d.)

Table 1: My own adapted table of timeline of significant events within New Zealand Education system.

To understand the number of organisations that contribute at a high level to the conversations on inclusive education in New Zealand see figure 5. This diagram gives a tip of the iceberg understanding of the organisations that are involved; however, there will be many more who also play a significant part in decisions that are made for vulnerable young people in New Zealand.



*Figure 5: My own - Organisations and Legislation in support of people with disabilities.* 

In bringing the literature in relation to the United Nations Conventions and the National Legislation and how it contributes to my research to a conclusion, I will now discuss the importance of lived experiences by drawing on research that examines both advocacy and empowerment.

#### 2.3 Lived Experiences – Advocacy and Empowerment

The fundamentals of advocacy and recognising the rights of individuals is stated within the founding document of the United Nations Convention on the Rights of a Person with a Disability (*Protocol - Convention on the Rights of Persons with Disabilities and Optional Protocol*  *United Nations*, n.d.). Moriarity and Dew (2011) examined the participation and advocacy of a person with disability during the UN Convention from the perspective of a person with a disability. The opportunity to advocate as a person with a disability on the world stage was seen as positive with their contribution being received by all as unanimously favorable.

Without the ability to be an advocate or self-advocate, individuals with disabilities are placed at high risk in terms of community participation (Heah, Case, McGuire, & Law, 2006). According to Dillon (2013), there are five steps to advocacy ranging from knowledge on rights through to the sharing of relevant information. Mithra (2009) agrees with this, thinking with a rights-based, person-centred, watching brief as a witness/observer. Used in an everyday setting of a school environment, the advocacy skills of a teacher are seen as able to achieve inclusion which is multi-layered. According to Petrie (2018), a combination of creativity, patience and reflective practice builds the foundations of a strong advocate and advocacy relationship (Petrie, Devcich, & Fitzgerald, 2018).

In discovering the world of assistive technology, we see the empowerment opportunities, with the right support, being endless. Creating independence for individuals with high and complex support needs with assistive technology is incredibly empowering (Dalton & Hoyt-Hallett, 2013). Empowerment can be achieved through a variety of interventions and methods. It is well established that children with disabilities are at risk of limited participation opportunities (Dalton & Hoyt-Hallett, 2013, Healey-Ogden & Austin, n.d.). However, according to Townsend et al, the promotion of assistive technology can be "enabling and inclusive". Studies carried out in a number of different environments, across a variety of ages and demographics, state that assistive technology is both welcomed and necessary in generating a sense of engagement and independence (Beyer & Perry, 2013).

Advocacy and Empowerment are both actions and outcomes respectively. To advocate in a meaningful way will produce empowering opportunities for individuals as well as having an impact on the community in which an individual is engaged.

The following section looks at both the medical model of disability and the social model of disability.

27

#### 2.4 Medical Model of Disability

In drawing on literature regarding the medical model of disability, I see it has been described as a framework that is medicalized (Ba, 1990, Coles, 2001) and places the impairment or the issue of the disability with the individual. Ba (1990) argues that it is inappropriate to place disability within a medical model framework as he claims disability is a social state as opposed to something that needs to be cured. Oliver claims that the medical establishment is trained to diagnose and treat illnesses not to alleviate "social conditions or circumstances" (Ba, 1990, p.3). There is acknowledgement of tension between both the medical and disability establishments with one focusing on what has been described as the deficit model and the other as a model which draws attention to the barriers and obstacles that are created by society (Ba, 1990, Shakespeare, 2013). However, according to Fatoye at el (2018), neither the social model nor medical model of disability is effective if used alone. Figure 8 (below the definition box and over the page) gives a visual description of the differences between both the medical model and the social model and the social model.

The Medical Model of Disability according to the Oxford Dictionary of Human Resources " is the view that disability originates in a physical or mental condition that is intrinsic to the individual sufferer and who thereby experiences a reduced quality of life. The policy response that flows from this conception of disability is one that seeks to cure the underlying medical condition or alleviate its symptoms; for example, through the use of prosthetics or the prescription of drugs. Disability rights campaigners are often critical of the medical model of disability because it promotes a negative, disempowered image of people with disabilities. They tend to favor, instead, a social model of disability". (Heery & Noon, 2017b)

#### MEDICAL MODEL AND SOCIAL MODEL OF DISABILITY:



Figure 6: medical and social model diagram

("Medicine, Social, Green, transparent png image & amp; clipart free download," n.d.)

#### 2.5 Social Model of Disability

According to Oliver (2013), the concept of the social model of disability was developed from the Fundamentals Principals of Disability document which was first published in the 1976 by the Union of the Physically Impaired Against Segregation (UPIAS) (Oliver, 2013). The original statement from the UPIAS booklet says:

In our view, it is society which disables physically impaired people. Disability is something imposed on top of our impairments, by the way we are unnecessarily isolated and excluded from full participation in society. Disabled people are therefore an oppressed group in society. It follows from this analysis that having low incomes, for example, is only one aspect of our oppression. It is a consequence of our isolation and segregation, in every area of life, such as education, work, mobility, housing, etc. ("Fundamental Principles of Disability – National Disability Arts Collection & amp; Archive," n.d.).

In the early 80's, Oliver (1983) drew on this concept to demonstrate a clear definition between what he described as the medicalization or medical model of disability and the socialization of people with impairments. Oliver's aim was to create a framework that would be easily understood by not only his students but also by professionals. The social model of disability is referred to throughout literature by both proponents for it and critics against it. Oliver admits to

the social model being far from perfect, stating it was only meant to be a tool to help improve not divide people's lives (Ba, 1990). However, he also claims that with the support of human rights groups the social model of disability is gaining traction and challenging the thoughts of the medical establishment (Oliver, 2013). Shakespeare and Watson (2001) also appeal for the social model to be maintained as they too believe it is yet to be embraced fully by medical professionals as well as academics who are working within this discipline. Shakespeare (2013) claims that since the earlier academic work of Oliver (1990) among others there has been little development and goes on to state that there are a number of instances where the social model has become a hindrance to its own success.

The Social Model of Disability according to the Oxford Dictionary of Human Resources "is the view that social practices and attitudes render people disabled by making it difficult or impossible for those with physical or mental impairments to participate fully in social and economic life. The implication is that social practices, including those at the workplace, should be adjusted to allow the full inclusion of disabled people in social and economic activities. The social model of **disability** is often espoused by disability rights campaigners who wish to emphasize that the problems faced by the disabled arise as much in discrimination as they do physical or mental conditions. Under pressure from campaigners, it has begun to shape disability employment law, most notably in the requirement on employers to make reasonable adjustments to accommodate those with an impairment (Heery & Noon, 2017a).

In looking to understand both models alongside one another (See Table 2, below), I have used the work of Haegele ("Comparisons between the medical and social models of disability discourse. | Download Table," n.d.) to show how both models of practice work when presented with the same question.

In bringing both the medical model of practice and the social model of practice to a conclusion, I will now, in section 2.6, discuss the collaboration and the various models that are used across the relevant disciplines.

Торіс	Medical Model	Social Model
What is a disability?	An individual or medical phenomenon that results from impairments in body functions or structures; a deficiency or abnormality	A social construct that is imposed on top of impairments by society; a difference.
Targets of intervention.	Fixing the disability to the greatest extent possibly; normalizing.	Social or political change in an effort to decrease environment barriers and increase levels of understanding.
Perceptions toward individuals with disabilities.	The individual is faulty.	The individual is unique.
Perception of disability.	Being disabled is negative.	Being disabled is neither positive nor negative.
Effects on individuals who are typically functioning.	Society remains the same.	Society evolves to be more inclusive.
The agent remedy.	The professional	Can be the individual, an advocate, or anyone who positively affects the arrangements between individual and society.

#### 2.6 Collaboration

'Collaborative Practice' has been described by the World Health Organisation (WHO) as a way multiple health organisations can work alongside one another to promote the access of quality essential services and to support peoples participation in national health services among other key focus areas ("World Health Organisation - What we do," n.d.).

There are several models that are referred to within the sphere of collaboration or collaborative practice which are used readily within the disability sector and across several professional disciplines including both the medical and education sectors (Golom and Frank, 2018). Golom and Frank (2018) explain the differences across the multidisciplinary team, the interdisciplinary team and the interprofessional collaborative team as being hierarchical, less hierarchical but not equal to no hierarchy or territory; all members are equal. Golom and Frank (2018) explain that the Interprofessional Collaborative Practice (IPC) model is used within the health sector or health sector framework; the Interprofessional Education (IPE) model aligns itself within an education framework, and, finally, the interprofessional collaborative practice model is used across all communities. Although the movement to interprofessional collaborative practice is still in its infancy and there are challenges that have been recognised by Golom (2018)

as complex, an approach of "think globally, act locally" has been recommended (p. 10). For interprofessional collaborative practice to achieve the desired outcomes of teams working together to build better outcomes, Molyneux (2001), through her research on interprofessional teams identified three key components:

- 1) The desire to be part of a collaboration with a willingness to share
- 2) Communication within the team,
- 3) Development of creative working methods.

A framework developed by Mentis et al (2001) and Zuckerman, Gal-Oz, Tamir, & Kopelman-Rubin (2015), aptly titled *Mawhai*, meaning 'web' and 'net,' has also described the interprofessional relationship as being "when professionals from different areas learn with, from and about each other interprofessionally. They develop a shared understanding of their different areas which facilitates their practicing together" (Mentis, 2001, p. 14). This description is was also used by Golem and Frank (2018) in *The Journey to Interprofessional Collaborative Practice – Are we there yet?*. Furthermore, Mentis et al (2001) goes on to say "Learning together resulted in increased confidence and competence to practice more effectively together" (p.74).

A relatively newly-coined model with the world of collaborative practice is that of community conversations. The point of difference that community conversations have in comparison to the IPC (health model), the IPE (education model) and the IPCP (community model) is that it brings all stakeholders together with a set structure in the way conversations are hosted. The community conversation model was developed from the world café model which focused on lessons in dialogue has been described "as an approach that provides a structured way of generating and prioritizing local solutions that align with a community's cultural priority and available resources" (Carter & Bumble, 2018, p. 13). In looking at the world café and the community conversation being more aligned with the social element of society. Both models, however, are working towards drawing in resources for the betterment of a situation. The literature on the community conversation model of practice can be used across several environments. The focus to date sees this model of practice being used as a way to

generate change at a policy level (Flippo & Butterworth, 2018) as well as across an individual's support networks. Both are examples of two different audiences using the same model. Carter and Bumble (2018) explain in The Promise and Possibilities of Community Conversations: Expanding Opportunities for People with Disabilities, that although the practice of community conversation within the disability sector is relatively new opportunities are being created (Carter and Bumble, 2016). The sustainability of these opportunities has not been researched further however (Carter & Bumble, 2018, Trainor, 2018).

There is currently discussion around the use of 'community conversation' as a form of methodology. Carter has stated that community conversation is a unique methodology, but Trainor (2018) has countered with the idea that community conversation is not an established research methodology yet could be used as an integrated part of research projects. In building an understanding of how community conversation are practiced the benefits have been described throughout the literature as positive (Carter & Bumble, 2018, Trainor, 2018, Parker-Katz et al., 2018, Flippo & Butterworth, 2018). The research to date on the challenges is unknown but sustainability has been highlighted as a possible challenge (Parker-Katz et al, 2018).

The below figure (Figure 9, over the page) presents more ideas on that can be built upon using the community conversation approach.

In concluding this section on collaboration, I will now discuss both the benefits and challenges that lie within the world of assistive technology and how collaboration sits within this space.

33

Figure 7: My own figure with additional ideas on the benefits of community conversations.



#### 2.7 Technology in an unspoken world



Figure 8; Assistive technology; image from Assistive technology Facebook page.

Assistive technology for high and complex users is becoming more and more advanced with tailored software being designed with a specific end user in mind. The spectrum of systems and software available is huge. One example, described by Batorowicz (2016), is a form of communication software called augmented alternative communication (AAC) that is becoming so finely tuned it can now be programmed to interpret utterances (Batorowicz, Mcdougall, & Shepherd, 2006). According to Wallace et al, "using both technology and strategic approaches, AAC often focuses on the restoration of cognitive-communication and/or physical speech impediments" (p. 43). Natural Language Generator is another assistive technology communication tool that can be implemented to create a story telling platform. Tintarev (2016) claims that a challenge with this software; however, is that the options for sentence construction is only as good as the pre-programmed library. If there are limited options, it can be potentially frustrating for the individual engaging with it (Tintarev, 2016). For ease of navigating a system or understanding what is required, Tintarev (2016) suggests that assistive technology is divided into

subsets depending on what form of software is being created. The subsets suggested are sensor, environment, promoting and training, communication and telecare.

According to Shih (2012) "it is very important to provide collaborative working/learning opportunities in computer operation" (p. 10). Shih's research looked at relatively low-cost software modification and how individuals could use a mouse scroll wheel in a collaborative setting to communicate. Although this research and design only allows for collaborative pointing it is an option that should be considered. The mouse scroll wheel allowed for collaborative conversation between parties (Ching Hsiang Shih, n.d.). Furthermore, Hamel states that "simple devices such as a pager or a remote for the tv can have a significant impact on people's lives in that it enables them to have a sense of control" (Hammel, n.d., p. 57).

The benefits across all areas of accessibility and the power behind accessible software was highlighted throughout the literature. To be able to create a personal narrative and share thoughts through having tailored accessible functionality is galvanizing for all parties (Tintarev, 2016, Ching Hsiang Shih, n.d., Katsioloudis & Jones, 2013). According to Katsioloudis (2013), having an awareness of the benefits of assistive technology, learning about the research in a sometimes-clinical setting to truly understanding the implications of it in an individual's life certainly makes the subject very real. Hamel (2003) shared two research samples from both an assistive technology end user and a community support worker. The end user stated that

"Simple, I talk. I need it. When broke, I no talk" (p. 331).

This is followed by an extract from a conversation with a community worker in sharing that -

"He's all over the place with his new chair. Before that he never left his room unless someone had the time to roll him over to another room. Now he's whizzing down the hall and around the block. I can't keep track of him sometime. One day they were tearing up the sidewalk outside the house and he couldn't use the new chair. It was like he was a different person again. Dependent on people to move him everywhere. It's kind of amazing to think about – with it he moves, without it he's dependent on me (Hammel, n.d., p. 331)

36
Being able to access assistive technology when required is a challenge. Also having ongoing support once it is accessed is also a challenge (Hamel 2003). Through the research of Lancioni (2016) it was found that people with multiple disabilities can have serious difficulties managing their independence including both activities and communication, particularly when support people were not immediately available (Coleman, 2003, Lancioni et al., 2016, Hammel 2003, n.d. Seelman (1993) claimed through his research on *Assistive Technology Policy: A Road to Independence for Individuals with Disabilities* that the equity among people with disabilities and those of the able -bodied population were impeded. It could be argued that 26 years on the same challenges are still occurring in terms of equity and access to resources (Seelman, 1993).

In looking at the issues around the resourcing of assistive technology, slightly offshore from New Zealand, Ratliffe (2012) looked at the challenges within the Pacific Islands. Ratliffe's (2012) research found that the challenges in accessing appropriate technology was challenging although is gaining access to western media (Ratliffe, Rao, Skouge, & Peter, 2012). The difficulties seem to be two-fold in that not only accessing assistive technology is a challenge but also in knowing what assistive technology is available and an individual's rights to that (Ratliffe et al., 2012).

To achieve the development goals of building assistive technology resources that are tailored specifically for individuals with high and complex support needs information is required to be housed in a cloud-based system. The following section discusses the storing and sharing of such information.

## 2.8 Storing and Sharing Information

The final area of interest within the literature review is the storing and sharing of personal information. For the purposes of this research it is acknowledged that the intricacies, algorithms and other such things of security in a cloud-based storage system fall outside the scope of the research question. Security is a concern for all of us regardless of our executive functioning ability. For a growing portion of the population more and more of our personal data is being

transitioned onto a cloud-based storage system. Being able to access our health information through an electronic health service, our finances through an online banking platform and education records through our education provider are to name a few. By engaging in such services, we are assuming that our information is secure.

With cloud-based storage one of the first questions that is often expressed is around security. In acknowledging this as a real concern by both Wilson, (2011) and Yaksel (2016) it was emphasized that firstly all aspects of storing and sharing information are taken very seriously, and systems are becoming more and more difficult to penetrate. Wilson (2011) has confirmed that although the security system is not flawless however "the overriding benefits from cloud computing are such that businesses could face real challenges in the future if they resist adoption" (p. 99). Yuksel (2017) agrees and states that although having access to cloud-based information has numerous advantages there are also ongoing concerns in terms of privacy, security and integrity of hosting information (Yukel, 2017). The benefits in the transitioning of information begin to emerge over a persons' lifetime. Petroulakis (2017) claims that 'life logging' on some devices is proving a challenge due to memory, resources and computer power (Petroulakis, 2017).

Conceptualizing and having the opportunity to trial the concept in a testing environment is hugely beneficial (Zuckerman et al., 2015, Miura et al., 2012). The insights gained help set the scene for further development if there is in fact a positive return on the concept. The importance of trialing a concept has been demonstrated both through the Tangi Plan (Zuckerman et al., 2015) prototype and the BarrierFree (Miura et al., 2012) website pilot concept. Zuckerman et al were able to reinforce the importance of testing. Tangiplan gathered feedback from their participant group and although small in numbers they were able to gain some sound insights. Tangiplan identified two problems which were the stability of the internet connection and battery life of the devices. These two problems are very generic in nature. Both Tangiplan (Zucket et al., 2015) and BarrierFree (Miura et al., 2012) have both implemented accessibility functions and continue to develop their platforms.

The following is a summary of my literature review where I will discuss briefly the eight areas of literature that I have engaged that have assisted in building my understanding of the intricacies of each of the disciplines.

## 2.9 Summary of Literature Review

In this chapter I have drawn on literature from across a variety of disciplines which inform the research question: Research the concept of a storing and sharing website for individuals with high and complex support needs. Literature selected from the area of Advocating for Individuals with High and Complex Needs, Lived Experiences and Empowerment, Understanding Interprofessional Collaboration, Using Technology in an Unspoken World and Information Storing and Sharing Websites share the solution focus approach in looking for ways to support people with high and complex support needs better.

Due to the wide range of literature covered within this review, I reinforce the importance of the community conversation model that takes into consideration a wide variety of sectors and disciplines as well as legislations and policy to help inform and give guidance where appropriate (Flippo & Butterworth, 2018). It could even be suggested that the World Café model may fit this concept as it has far wider reach (Carter & Bumble, 2018).

This literature review has provided a wide-angle lens where the document of the United Nations Convention on the Rights of Persons with Disabilities has been outlined followed by an overview of the New Zealand Education system and a timeline of the journey of 'special education'. By looking at the wider landscape of disability and understanding the founding documents and legislation at both an international and national level, we can begin to plot a course towards understanding our current situation.

Investigating the literature of lived experiences and the benefits of advocacy and empowerment, I have been able to gain a deeper appreciation of both the challenges that are presented for individuals as well as the importance of continuing to advocate for people who are unable to represent themselves. The flow-on effect of positive advocacy is evident in the literature on empowerment and how this can impact the holistic health of an individual.

Each area of literature presented through my review has been a key in bringing together a much greater understanding of the wider world. Each discipline has its own complexities and idiosyncrasies and by gaining an understanding of the work that has gone before begins to build an important foundation of the landscape. Both the medical model and the social model of disability literature discussed in my review has brought a greater understanding of the historical context to both. What might have been described as black and white thinking within this discipline has now moved into a shade of grey where an understanding of the importance of both is now clearer for me as a research practitioner.

Collaboration both internally and across partner organisations is discussed with the benefits and challenges being brought to attention. The academic literature describes what different forms of collaboration look like within various organisations yet also acknowledges the outcome in achieving this remains elusive. Collaborative practice on paper and within organisational strategies looks quite different to what is occurring within our communities.

There is an ever-increasing world of technology being developed with access to information technology resources becoming more readily available. In this review, I have mentioned some examples of resources that are available for individuals with high and complex support needs as well as sharing examples of technology and how it was piloted in the development phase. In developing technology for individuals with specific needs certain aspects must be more considered for the resource to be fully accessible. It is an exciting time in the space of both disability and information technology as the world of virtual interaction, robotics, and augmented assistive technology becomes a normal part of our day to day world.

The storing and sharing of information are the key concept of this research project. The main area of concern that was shared regarding this was the security of such information. The literature however has identified that concerns around the security of information is not a valid reason not to pursue placing information into a cloud-based system that others could potentially have access to. The literature has suggested that no system is 100% fault proof however the technology is getter better with security being unlikely to be penetrated.

# **3 CHAPTER THREE: Methodology and Method**

## 3.1 Action Research Methodology

Jean McNiff (2017) describes the action research methodology as an enquiry conducted by self into the self. As a practitioner you think about your own life work as well as asking why you do the things you do, and why you react the way you do (McNiff, Edvardsen, & Steinholt, 2018). The rationale in selecting Action Research as the methodology of choice is the relevance it has in guiding and advancing me as a professional practitioner as well as aligning itself to my research project. Action research methodology has enabled me to be both researcher and reflective practitioner as well as enabling me to work in a co-operative partnership with the research participants to look for solutions to the identified problems.

The steps taken (Figure 11) in the process of action research take a methodical path that supports the building of both evidence and data towards the research project.



Figure 9: The action research cycle.

(Adapted from: Azhar, Ahmad, & Sein, 2010)

The first step of six is in identifying the problem or selecting an area of focus. This step is vitally important as it is the first step in assembling what will become the project. The main element in step one is what do I want to investigate? For my research project I wanted to

investigate if there was a way information could be shared across an individual support networks therefore, I was looking to test the concept of a website that would promote this.

The second action within this model was to identify a solution to the problem. The solution I identified was to potentially create a platform that would host relevant information that was easily accessible for stakeholders working alongside an individual. The solution would hold key pieces of information that helped inform and educate those working alongside individuals and their families.

Step three is to try out the solution in such a way that information can be gathered and considered. Step three for my research project involved sharing the concept of a website with 16 individuals across 5 different demographics. The concept of the site was presented with each interview participant sharing their thoughts on both the benefits and challenges they believed were relevant to them or their profession.

Steps four and five are the gathering and analysing of the data. For my research project interviews were recorded and transcripts transcribed. I coded the themes that emerged from the data into an excel spreadsheet by sectioning off each question then placing the relevant themes into each question. Each theme was colour coded to help in the identification.

The final stage of the action research methodology model is the writing of the report. For this part of the process I placed the information within a qualitative framework that allowed me to speak to the data using the voices of the research participants.

#### 3.2 Qualitative Vs Quantitative

I have chosen to use a qualitative research approach within this research project as opposed to a quantitative research approach. In doing this, I have been able to observe and interpret my findings using a small research participant group rather than a much larger sample size. The qualitative research approach using the action research methodology has given myself as the research practitioner the opportunity to work intimately with the participants rather than being uninvolved and working at a distance. Although qualitative research has brought a certain amount of subjectivity to the discussion, I believe for the purposes of my research question, this approach was the most appropriate as the voices of the participants have been crucial in gaining an understanding as well as giving me valuable guidance of the situation. Being able to explore questions that have a 'why' value attached to them has been of value within this context. Due to the nature of the research and the demographics, i.e. being with disabilities there has also been value in face to face interviews rather than an online questionnaire.



("Strengths and weaknesses of Quantitative and Qualitative Research | WeeTech Solution Pvt Ltd," n.d.) Figure 10: Image of quantitative vs qualitative data

According to Weetech Solutions, the pros and cons of quantitative versus qualitative data collection depends on what is looking to be achieved. For the purposes of this research project I was looking to explore an issue and gain an understanding to a problem. Therefore, with this in mind the qualitative approach is the correct one to place this research within. Qualitative and quantitative research differ with qualitative research for example focusing on the analysis of large numbers and measurements and qualitative data looking to capture the details of descriptions and observations (see Figure 12).

## 3.3 Research Method

The method I have used to collect data, analysis the data and report on the data is in a qualitative form. The steps I have taken are as follows.

- Ethical approval was granted by the Otago Polytechnic to carry out the research. (see appendices).
- Flyers (see figure 13) were created and distributed through the Otago Branch of CCS Disability Action (see appendices).



Figure 11: Flyer recruiting interview participants.

- Information did begin circulating eventually after a phone call was made to CCS.
- Participants made direct contact with me and interview times were arranged.
- Interviews were carried out at either Sport Otago (boardroom) or at the research participant's home.
- The interview participants signed the consent forms and gave permission for the interview to be recorded.
- Each interview took approximately 40 minutes.
- All interviews were recorded.

- Each participant was shown the concept of the website through screenshots of each page (see figures 5 through to 12) that I had bound into a booklet. This was originally a backup plan but became plan A due to ease.
- The interviews were transcribed.
- I explored the use of NVivo to analyse the data however due to the amount of data collected, I concluded I could create a system in a shorter amount of time that would achieve the same or a similar outcome. Table 3 shows the steps taken to group the data in the questions into themes.

	Teacher - AF	Student - AK	Support Worker - AR	Teacher - Beth Downie	Parent - CM	Parent - DM	Support worker - I	Med Prof - IC	Med prof - JB	Parent - JK	Student - KD	Teacher - NB	Parent/student - RB	Med Prof - SG	Support Worker - TM	Med Prof - TW
fhat are you initial npressions?		Good concept/helpful to have stored in one place instead of all over the place where things get lost.	Fantastic concepttrailed off with ideas.	A very successful tool/ everything is just bits of paper/pear shaped/all in one jacef doesn't look complicated/in one central placefno brainer/ho kind of flow on.	about is on control/I don't think there is anything ese out there/good broad range of stuff/	Could see how it could wrk in the education setting/ every year we lose information/have to reshare information all the time/information doesn't get shared/ without me having to the the medium all the time.	I think it is extremely valuable/record of the person/ see their strengths and weaknesses/thing definitely lacking for me as a TA/information isn't passed on/find out last minute/one stop shop.	stand I would be invited at times and not at other times.	idea/information should be held with the child and parent/should be albe to selct who knows what/positive	The concept is a really good one/informatin already provided/lost information/one system/SENCO has left/four different places that have different vags of recording/makes life easier - heaps easier.	The concept is well designed and allows persons to easily access their collated data/streamline information/contr ols for allowing certain percle to access certain parts of site.	I'm very impressed/consi stently wise - going forward for the child is very beneficial/ streamlining It/ going to be really productive.	I think it looks quite good actually bits of information everywhere? I look back on old diaries, spreadsheetstr acking activities.	I'm very impressed/a long time ago multidisplinary teams who did share infromation/4/5 different professionals and none of them meet/fills a gap/person controls access to it/fills gap of storing it securely.	I think it quite good because it gets those professionals talking to each or here as required or neededfrecord of portfolio for kidsf seesaw programme.	Llike the concept of it because often people art left having to reiterate all their information lots of times so the overall concept is good/ seems straight forward
	increase workload/a															
o you believe the		Yes/There are	Yes/Its vital if x is		Yes/It becomes	Yest being able	Yes/get ideas on		Yes/fish hook -	Yes/ gain	Yesleasy	Yes/collaboratio n/ different	Yest sharing	Yes/person with	Yes/professional	Yes/information
oncept of a website here information	service/big teams/not a	often a lot of support	unable to be contacted we	central place/no brainer	the one piece of information or	to hve access to stuff about	how to implement	is lost in translation/hadn't	time/parents in control which is a	information/read up and gain	access/made available/ideas	nf different expertise.	information/ education/things	disability has the control/first time	s change and leave/new people	is siloed/have to reinterate
ould be both stored	person to help	people/they do	need to act in the		database that	herself.	strategies/this	got the full		information	lost in		get lost/so much	for me a	come inflook	information/not
nd shared among	negotiate	not need to	moment/ we		people usełit		worked really	understanding/im		before they start	trnslation/not		to learn/hard to	repository of	back on	having to
rofessionals would	through that	repeat	need to know		becomes the		well/if you can	portant to be able		trying to work	repeating		remember	information.	history/what	duplicate.
e helpful for both the		information if a	who does		place people		get one idea out	to		with whoever.	information/acce		everything/peopl		happened/whats	
ndividual and their	people/communi	new support	what/it's all in x		know they need to		of it its' one idea	capture/incrediab			ss to important		e to know about		been tried/good	
upport teams? If yes hy? If not.why not?	cation out could be really	person cornes along.	head/all the little things you take		go to to find out information/com		you didn't have before.	le amount that become apparent			information.		his interests/not usre how		tohave all in one place.	
nge ir noc,wny noce	successful/it	along.	for granted/what		munication in		berore.	about a young					effective		piace.	
	can't become		provides quality		school/it gives			perosn as					communication			
	owned by one of		of lifer a central		her a			theyy're growing					is between health			
	these single		place to be able		voice/comms			up, it's really					professionals.			
	entities/advocati		to find		book some			important to					-			
	ng for		information.		people don't read			keep track of.								
o you believe there		Manage our own		Transition/ideas	Continuity among		Different	Limited ideas	Our system	Personal	Control over	Good having	Information gets	telling story over	Easier for people	Background
ould be any positive autoomes from having	ication in huge teams.	profiles/not having to repeat	staff/little things are important	are embedded/you'v	professionals/ev eryone can	information the better/cruing out	ideas/seeing what works/time	about what is possible/transitio	doesn't have a lot of good stuff	approach/knowin	information/sens e of	information stored.	lost/better understanding of	and over.	to talk to each other/anything	information/ if I known
ccess to a vebsite	cearns.	naving to repeat stuff.	when you can't	e had five years	eryone can access it/not	for information.	poorłyou don't	possiblertransitio	iot or good stuff	g background/not	e or freedom/qualtiy	Stored.	a persons needs.		otherranything that adds to	that/personal
uch as this?		sturr.	manage the big	of creating/don't	reliant on us	ror information.	get planning time.	system is left		repeating stuff.	of life/support		a persons needs.		communication	attributes/build
			things/life is much richer/see	what they do or	articulating that all the time/capturing			behind/be really specific about			team can access informatin					rapport.
			them as a social	how they do it/no				skills/carry			quickly.					
			beings ,not a	flow on through.				forwardfall			1					
			medical					possibilities and								
			beings/emotional					skills and								
			connections/					interests that								
								might have								
								become apparent								
								at one time early								
	1		1	1	1			on and maybe			1		1			1

	Feacher - AF	Student-AK	Support Worker- AR	Teacher-Both Daunie	Parant-CM	Parent-DM	Support worker - D	Mod Prof-10	Modprof-JB	Parant-JK	Student-KD	Teacher-NB	Parent/student- RB	Mod Prof-SG	Support Worker - TM	Med Prof - TW
	Incluing a neadflite rama thing use dan's currently avoid lastic a car y to avoid lastic a car y to avoid a start and and avoid a start a dan da raning the amount of raning the complete according down to burd down to burd down to burd avoid to burd down to burd dow	Gand canceptholpfulta havertardia and place intradia di all aver the place uhere thingrast lart.	Fontartic conceptd.strailed off uith idear.	A very russerful taal verything ir jurbing of paper foar rhap of all in ano place flaar in ano ano control place flaar in ano place flaar in ano place flaar in ano flau an.	Well Have interested resources the perron the strict down't an controll don't think there is anything also out there is anythere anythere is a strict of the anythere is a	Could rook how it could work in the ducation rottine of overy your to lare information flower to reach are information all the mediatemation down 'get hared ut have the medium all the time.	Ishinkini extremely valuable for eard at the person free the person free the person free definitely lacking for mear a Takindarmatian inn's passed and find aut lack minute fane stap shap.	Simple and logical/real/read sarily understand/under tand logid at time and net at table timer.	I think it's a fantartic ideofindarmation, should be hold with the child and use of the know whether the show whether the show of the show whether the show and the show how they do it got lart firth hook - time.	The conception really and ane finfamatin already provide dilart information dans rystem SENCO has leftfour different usys affront usys af recordingtmaker life easier the app easier.	The concept is well designed and allow porrar to ourly accert to ourly accert and data/streamline information/const- our for allowing certain peraleta accert ontain partrofisite.	In very improved fearviet ently use a future child in very bonoficial streamlining if gaing the bonolly praductive.	I bhink is lamke quite quand actually fibre of information avery under all lamk back an all diarier, aproad the atutra aking activities.	I'm very improzed a lang time aga multidig linary team wha did schare inframation ddS different professional and professional and them meet fill a geoperson cantrals accers to iteffill ago of staring its scorely.	I think it quite quad because it get that o preferrian de taking to each ather are quited ather are quited ather are quited perfains far kider zeo gu pragrammo.	Hike the cane of of it because of people are left howing the relief all their information their times and the second second canepe bit quasi second second second for used.
n ynu believe the cuncept of a uebrite here information could be both stared afskord amna professionale uuld e belgfol fur both the individual and heir support team? If yer uby? If at,uky not?	uarklaad saringla uary fa doucating far yourself. Carl Wrap around toomrinat a oorran to help negatisto through hat many ooplef ammunic atinn out could be really ruccarfulfit can't cocarfulfit can't cocarfulfit can't cocarfulfit can't cocarfulfit can't an of theoreringlo notificer advocating for themrelver	YorfThoro aro afton a lat af ruppart peoplethey da nat need ta repeat infarmation if a neuropart for an autor a lang.	Yearthy vitalify sin unable to be enhaded a unable to activities memory us need to find a unable deer headfall the little and the little gravite data and gravite data and provide a quality of lited a control place to be able to find information.	fuhat provider quality af lifel a contralplace to be able to find information.	Yarfit becomer the ane piece of information or databars that people workt becomer the piece papels know they need to get at the find out information feature aution that find out autor that information feature autor that the piece bankrame people dan't second it flant's how second it flant's	Yorf being able to hvo accers torstoff about hereolf.	Yorfqotidoorn hauta implement stratogiorthlir uurkod rodhir uurkod rodhi	Year finformation in lart in translation/headn'i ant the full understanding/imp artant to be able to capture/incrediable came apparent about a young person as the yyy're growing up, it's really important to keep track of.	Yarffirh hawk - timorfparanta in cantral which ir a quad thing.	Yorfqain informationfroad up and gain information before they start trying to work with wheever.	Vorloary accortmado availabletidar Iartin translatian/nat repeting information/accor rtaingertant information.	Yorfcallabarationf difforent expertire.	Vertzhering information/ education/thinach education/thinach learn/herd to remember overything/spepile to know about hir inforcertrenature hau offective cammunication ir between health profeszionals.	Yarfporran uith dirability har the cantral/first time far me a reparitary of information.	Verfprofessionals change and leaverhous people amme influents back an historyfuhat hapeene dfuhats been tried grand have all in one place.	Verfinformatio rilsed/have to reinterate information/ne having to duplicate.
r ynu bollovo thoro unuld bo eny ritioo nutemnor fram kaving accosr a uobrito sueb ar thir?	Sharinqf communi cation in huqo loamr.	Manaqo murmun prafilor frat havina taropoatztuff.	High turnsvor af rraffHitle thing are impartent uhen ywu can't manage the big thing Hiffs ir much richerfree them ar arsoial being rom tional canno chiant canno chiant	Transitionfidear are ambeddedfynu'we had fiwe ysarr af areatingddau reatingddau haw the y an articular haw the y da it/na fiau an thrawgh.	Terretunde cume cume professional rever yone can account infant reliants na w articulating that all the time/capturing liferpan.	Maro information the betterforging autfor information.	Different ideatfooing uhat uarlorftime paartynu dan't got planning time.	Limited idear abart that ir parifield traction fin the current system it left behind the really regarding abart prodifies abart providing the second chiller day the second chiller and interact parifield that became apparent an and maybe used a curlity be foregate.	Oursystem deen't have a let of good stuff	Porzenal appresch/knawing backgrawna/net ropoatingstuff.	Cantral aver infarmatianfronre af free dam/quality af life/suppart team can accar infarmatin quickly.	Gaed hoving information stored.	Information, gote lastido estar understanding af a persone needs:	tellingstary over and over.	Earlor far people to tails to each ather fan ything that addr to cammunication	Background informationfi known thatfporrand attributorfbui rapport.
qativo uutcumos fram Laving accoss a uobrito such ar this?	minimizing uarklaad nat adding ta troxpoctationzfor sating quidolinozfparontz	Accorre information/infor mation gotting dolotod.	Perronal intentiyou'll alwayr get people who have negativity towardr ittred tape.	Noodr to be really urer friendly/quick and eary/minute anything ir too hard, too complexbig	tao much offort no one urer it/all traffic forcer people to ure it/X ounr it/all offort being put in and it	Funding it a thors according information.	Giving permission/not having access to device at school.	athorwire. Health information ir degrading/restrict ive about future/teachers and TAxee	Safo from hackorr/havo qood profilor torhou hou to uro it/privacy.	Privacy izruortziąn awaivor.	Parribly miruro of information/urong pooplo gaining accorr/gotting pooplo tartoro in ano place could be	Safoty of information/rostri ctions woro adhorod to.	Porsonal information romains privato/privacy.	Canfidontailfpriv acy and canfidontial.	Internet goer down/ree it ar another job to do.	Socurity.

P13 P3		F0 1
Luey Rose	Rose Blanch Sophia	Dorothy I
Parent/student - Suppor RB AR	C Med prof - JB Med Prof - SG Med Pro	Med Prof - IC
	2 Medplor-38 MedPlor-38 MedPlo	Med PTOP-TC
1 1	should be held	
	with the child and	
	parent/should be	,
	albe to selct who	
	knows	
	what/positive	
	information	i
+	in the second se	/l would easily
		understand/under
1 1		
		stand I would be
		invited at times
	ther seems s	and not at other
	forward.	times.
bits of		
information		
everywhere/ I		
look back on old	4/5 different	
diaries,	what they do, professionals	
spreadsheetstr	how they do it and none of	
acking activities.	gets lost them meet	9
1 1	person controls	
	access to it/fills	
s	gap of storing	
	one coherent information and	
	website sharing it	
	a long time ago	
	multidisplinary	
	teams who did	
	share	
+		
1 1		
+	fish hook - time.	
1 1		
t	two   Question three   Question four   Question five   Question six   Quest (+) : [	two Question three Question four Question five Question six Quest

Table 3: Example of spreadsheet showing how information was captured.

- Once I had all the interviews transcribed, I read through them twice before I began my anaylsis.
- On the third reading I could see points of interest and common language begin to emerge.
- I created an excel document and headed each tab up with the research question.
- I then listed all the nine themes down the left-hand margin so as I was going over the transcripts if a theme happened to be mentioned in a question that it wasn't directly related too, I could still capture it. (See appendices).

# 3.4 Concept screen shots of website



Figure 12: Landing page



Figure 15: Logging in page



#### Figure 13: Administrators page

Role: Learner	My Entries
Bridget Meyer's Waka Alicia Kapa's Waka	Sport, Recreation, Leisure and Hobbies
w My Entries	You have selected the Sport, Recreation, Leisure and Hobbies topic. You can choose another topic by clicking the icons below.
	Once you have selected the topic you want, click "Create new" and type in your name and organisation to begin a new entry on that topic which you will be able to share with other people.
	If you have already added entries on this topic or if someone else has shared one of their entries with you, you will see these below. Simply click on an entry to open it.
	🔰 🍈 🎬 🛫 🐶 🔇 🥢
	Create new Fishing Inform® Tablers 16-0n-2018 Sport and Rec@ Tablers 16-0n-2018 Summing @ Statem 06-Me-2018 Statem 06-Me-2018

Figure 14: Options of six places to house information.

#### **Record the Details**

Below, add a title and date to the entry and any information you want on the subject. You can also add a supporting document - a Word or PDF file or an image or video.

If you want to add additional pages of information, use the 'Add' button at the base of the page.

When you're finished, click the arrow to the right of the page or the 'All Done' button to move

Entry title	Date	Supporting document
Fishing spots	October 2018	200
information		
awesome fishing spot		
		Browse

Figure 15: Window to house information.



*Figure 16: List of people within person's circle of support.* 



*Figure 17: Demonstration on how to drag and drop into appropriate file.* 



Figure 18: Wizard to say information has been shared.

# 3.5 Research Participants

The decision to use these specific demographics or interest groups was based on ensuring that I had diverse information from relevant stakeholders. The groups selected were individuals with high and complex support needs, parents, support workers, teachers and medical professionals.

Five Participant groups:

- 1) Individuals
- 2) Parents
- 3) Teachers
- 4) Support Staff
- 5) Medical Professionals



# research participants

**Call out for** 

Figure 19: Image for research questions.

("Seeking Research Participants? - Graduate and Postdoctoral Studies - Simon Fraser University," n.d.)

## Group One:

The three individuals were made up of one 22-year-old male, one 20-year-old female and one 12-year-old male. Both the 20-year-old and the 12-year-old were nonverbal and used communication devices to share their thoughts. The 12-year-olds' mother also helped with interpreting the information being shared. The two 20-year-olds were both studying at university and were able to grasp the concept with ease.

#### Group Two:

Two out of the three parents were local families with one being from out of Dunedin. The young people they supported were in intermediate, high school and tertiary education. All the parents that were interviewed were the mothers.

## **Group Three:**

Two of the support workers were female with one male. Two worked alongside the young people they supported in the school environment and the other worked within the home environment. The two that worked with the school environment had slightly different challenges than the one who worked with the home environment.

#### **Group Four:**

All three teachers worked at different schools throughout Dunedin. All were female.

#### **Group Five:**

Of the medical professionals, three were female and one was male. Two worked for a government organisation; one was recently retired, and the fourth was in private practice. The four medical professionals consisted of three physiologist and a physiotherapist. All were employed by a government agency.

Pseudonyms are used throughout.

# 3.6 Questionnaire Enquiry

The research study emerged from a set of open questionnaire questions designed to capture the thoughts of the five participant groups.

The questions were conceived with the goal of drawing out information to help facilitate the answering of the research question.

Conversations were structured around these pivot questions, with a view to generate word- and meaning-based thematic data that could be analysed later.

- 1) What are your initial impressions of the website concept?
- 2) Do you believe the concept of a website where information could be both stored and shared among professionals would be something that would help both individuals and their support teams? If yes, why? If no, why not?
- 3) Do you believe there would be any positive outcomes from having access to a website such as this?
- 4) Do you believe there would be any negative outcomes to having access to a website such as this?
- 5) What do you believe will be the challenges I am likely to face?
- 6) In terms of functionality and accessibility of the site what do you believe needs to be taken into consideration? i.e. Larger font size, voice activated instructions, visual aids etc.
- 7) Any further suggestions that you think I need to take into consideration?

# 4 CHAPTER FOUR:

## 4.1 Introduction to Findings:

This chapter will discuss the research findings by introducing each of the themes that became evident from the interviews. The themes were generated from the research transcripts where I was looking for points of interest that became prevalent across all 16 research participants. Although the research group was relatively small, I am confident for the purposes of this research project the representation of opinions is accurate. The method I used to capture this information is discussed in the methods chapter of my thesis. I will mention here however that the reason for nine themes is that I felt they all were worthy of discussion as opposed to blending into one another. In hindsight, however, I would do this differently and I will discuss this more in the methods chapter also. Not all themes will have a heavily weighted discussion section.

The nine themes that emerged from were:

1) Reaction

Any reference made to the concept – good, bad, or otherwise.

# 2) Current Situation

Any reference made to their present circumstances.

3) Advocacy

Any reference made to an advocate, advocacy or support.

4) Empowerment

Any reference made to equality, enablement, inspiration, encouragement.

5) Collaboration

Any reference made to partnership, teamwork, relationship, alliance.

6) Resources

Any reference made to known resources that are like the research concept.

7) Accessibility

Any reference made to accessible ideas, convenience, approachability, user-friendliness etc.

8) Security

Any reference made to safety, privacy, well-being.

9) Challenges

Any references made to barriers.

#### 4.2 Reaction

#### **Student – Reaction theme:**

Of the student group, two were able to communicate independently and one communicated through his parent. I was excited to share the concept with the students as I knew they would be brutally honest with their feedback and come up with ideas I hadn't considered.

All three made a positive affirmation statements with one elaborating on his initial reaction by stating that "allowing the person to collate their own data was a good idea" (Duey). Two of the students used a form of voice assistance technology to share their answer with the third answering directly. All three grasped the concept immediately. Although each of the students had varying degrees of support requirements, they were all able to recognise the benefits for all students with high and complex support needs.

Person	Group	Quote
Huey	Student	"Good concept"
Duey	Student	"The concept is well designed and allows persons to easily access their collated data".
Luey	Student	"I think it looks quite good".

## **Parent – Reaction theme:**

The parents' overall reaction was positive. All three moved quickly into explaining what they currently do to retain and share information. All three parents alluded to being open to anything that is going to help make life easier. The concept was understood easily with no questions coming back once it had been explained.

In presenting the concept to the parents they, like the students, gave encouraging first impressions. There was a consensus among the parents expressing that "the concept is a really good one" along with "I think it extremely valuable" and lastly "well I love it, useful resource".

Person	Group	Quote
Fonzie	Parent	"Well I love it/useful resource"
Chachi	Parent	"I think it extremely valuable"
Joanie	Parent	"The concept is a really good one"

#### **Support Worker – Reaction theme:**

Although the lived experience of a Support Worker is not as intimate as a parent there is an investment present that must not be overlooked. A Support Worker is an entrusted member of the inner circle and due to the nature of the work and the high level of support required becomes a key component in helping to build meaningful interactions.

The support workers initial impressions were positive, and they expressed these with statements such as it was a "fantastic concept" and "I'm very impressed".

Person	Group	Quote
Daisy	Support Worker	"Fantastic concept"
Violet	Support Worker	"l'm very impressed"

## **Teacher and Medical Professionals – Reaction theme:**

The Teachers' and Medical Professionals' initial reactions to the concept was positive. Both groups were interested in what else was available in the very early stages of the interview process and also acknowledged that they thought there was a definite gap in the market. The initial impressions were very positive with the teachers stating it was a "no brainer and a very successful tool", along with the concept would be "meeting a need". The medical professionals reinforced the message declaring "I think it's a fantastic idea" and "I'm very impressed".

Person	Group	Quote
Ron	Teacher	"A very successful tool / No brainer"
Harry	Teacher	"Meeting a need"
Rose	Medical Professional	"I think it's a fantastic idea"
Blanche	Medical Professional	"I'm very impressed"

#### 4.3 Discussion - Reaction

In looking at the reaction across all participant groups, I was aware that the time to digest the concept was very short and this initial heavily weighted positive reaction might have been given more from an encouraging stance on their part to me, as opposed to did they really think it was a good idea. However, I did think it was important to get participants' immediate reaction. I was particularly interested in the reaction of the student group as the whole concept is designed with them at the forefront of my mind. If they had been perplexed within the first few minutes of the concept being explained I believe it may have been back to the drawing board.

The parent group were also excited by how this concept could make the day to day logistics of their lives more streamlined for all parties involved with their young people. Through the reaction theme the parents is the person who does the personal cares, provides transport and speaks and advocates on their behalf. The parents are pivotal in providing what many describe as 'a good life' and the scaffolding that brings an individual's world to life and without this scaffolding an individual's world can become very isolated.

The support workers described how little information they have access to yet spend many hours with the person they are supporting each day. To be able to learn what has been done previously, what has worked and what hasn't worked would be of huge value.

The teachers all indicated positive responses and were perplexed that such an idea was not already available. The teachers wanted to support the children in their classes as best they could and were open to ways in which information could be better shared across both the school, home and health networks. The enthusiasm and energy shared by all the teaching staff was high.

The medical professionals were positive about the concept although their lethargic conversation for learning something new or having to contribute more than they already do was noticeable in comparison to the teaching staff.

Based on the 16 interview participants the overall response to the concept of a storing and sharing website for people with high and complex support needs was extremely positive. The initial impression has helped form a baseline as to whether the concept is starting in a positive space or a deficit space. The initial impressions showed movement within a positive space particularly in the student participant group. The concept is potentially worth pursuing.

To understand the reaction and the testing of a concept in the early phases of the design stage and the learnings surrounding this, I have been looking at the work of a number of assistive technology websites and journal articles to attempt to understand the importance of the initial phase of testing and secondly to see if there are concepts with similarities to the platform I am proposing.

In 2015 a prototype by the name of TangiPlan (Zuckerman et al, 2015) was created and tested with the initial impressions gathered to see if this concept did in fact achieve what the app developers were looking for. The concept of TangiPlan was to support children with executive functioning challenges in helping to manage their morning routines. TangiPlan is a time management app used on a tablet that encourages the user to set various times to complete tasks. The guidelines that TangiPlan used to develop the prototype were developed alongside experts (Zuckerman et al). The aspirations of the developers was to help reduce tension by keeping the children on task throughout their morning as well as providing a form of "flexibility and autonomy" for the children using the app (Zuckerman et al., 2015).

There are two parts of TangiPlan research that are relevant. The first area of interest that is applicable is the desire of the app developers to find a tailored solution for a learner or user group. In this case, it is children with Attention Deficit Hyperactivity Disorder (ADHD). Although

there are many platforms of assistive technology that have been developed for people with executive functioning (EF) challenges such as TaskTracker, ProcedurePal and Basic Calender to help support the management of time, these platforms are not specifically designed for children with ADHD. The point of difference with TangiPlan is the breaking down of individual tasks and the tailoring of these particular tasks within each time period (Zuckerman et al, 2015).

The second part of this research that is significant is the aspiration to examine and understand the initial impression of the prototype as it was being used by children *in situ*. The research was looking to acquire some form of "validation" (Zuckerman et al) as to whether the prototype was achieving what the app developers had in mind. By testing the prototype in the very early stages of development allowed for the initial impressions and insights to be mustered as well as adjustments to be made to the design of the app. One of the challenges that was identified at the early phase of development was around the actual devices themselves. The battery life of the devices as well as the stability of the wifi were the two main challenges (Zuckerman et al, 2015).

Although the technology concepts are clearly different the alignment between both project goals is in the desire to create something for a specific user group as well as pursue feedback from the end user group. To be able to analyse a concept or a prototype from a very specific user group provides an invaluable road map in moving forward. The Tangiplan research emphasised the importance of the trial phase. To be able to use the prototype development in the children's home and speak with both the children and their parents gave the researchers a baseline to begin from. Interviewing the families both pre- and post-testing showed clear areas of improvement over the time Tangiplan was used. To have a form of assistive technology that is specific to an individual's requirements can benefit more than the individual themselves. From the Tangiplan research it can be concluded that with the correct form of intervention improvements can be made for both the individuals and those whom are supporting them.

## 4.4 Current Situation

## **Student – Current Situation:**

Once the participants had given their initial impressions of the concept, they all continued to elaborate further in describing their current situation. The current situation theme is looking at what people currently have access to in terms of how they store and share relevant information. All the students had an awareness of the fiscal and human resources that were required to ensure things ran smoothly and, similar to the parent group, were frustrated at the repetitive nature of conversations. The students identified the challenges of having to remember so much information and stated as such: "hard to remember everything, so much to learn" as well as questioning how well information is passed on between the various organisations – "how effective communication is between health professionals".

Person	Group	Quote
Huey	Student	"There are often a lot of support people"
Duey	Student	"Hard to remember everything / so much to learn / not sure how effective communication is between health professionals"
Luey	Student	"Ideas are lost in translation / not repeating information"

## **Parent – Current Situation:**

In looking across the data the parents were vocal in sharing their current situation and the impact this has on both themselves and their sons and daughters in terms of accessing assistive technology.

In presenting the concept to the parent group the current situation of information being lost was a dominant issue along with information not being shared. There was a sense of frustration among the parents and a sense that their time was not of value.

Person	Group	Quote
Chachi	Parent	"Information doesn't get shared / without me having to be the medium all the time"
Joanie	Parent	"SENCO has left / four different places have four different ways of recording"
Fonzie	Parent	"Comms book some people don't read it"

## **Teachers – Current Situation:**

The teachers' interpretation of the current situation for the students they support was backed up by what the students had also identified as problems within the current situation. There were also additional observations from the teachers in relation to the flow on of information and the coming and going of staff. The sharing of information across the network was described as "pear shaped" with "no kind of flow on". There was also concerns raised about the turnover of staff with "professionals change and leave, new people come in". Possibly the most important learnings with the current situation is from the teachers' statement of "it's something we don't currently have, it fills a gap".

Person	Group	Quote
Ron	Teacher	"Everything is pear shape / no kind of flow
		on"
Harry	Teacher	"Its something we don't currently have / fills
		a gap"
Hermione	Teacher	"Professionals change and leave / new
		people come in"

## **Support Workers – Current Situation:**

The support workers shared several frustrations about the current situation and the lack of support and training they receive to carry out their roles. There were a few statements made around planning time and the haste which is required due to the lack of planning. The information that is not shared or "it's all in X's head" as well as I "find out last minute" were clearly frustrations.

Person	Group	Quote
Daisy	Support Worker	"It's all in X's head / the little things you take for granted"
Lily	Support Worker	"You don't get any planning time / this is something lacking for me as a TA / find out last minute"

#### **Medical Professional – Current Situation:**

Like the support workers the medical professionals the shared similar frustrations in the amount of information that is not shared with gaps appearing in individual's care plans. The medical professionals identified that information is not presented in a positive light and very much sits within the medical model framework.

Person	Group	Quote
Rose	Medical Professional	What they do and how they do it gets lost.
Blanch	Medical Professional	4 or 5 different professionals are working
		alongside somebody and none of them meet.
Dorothy	Medical Professional	A long time ago there were multidisciplinary
		teams.

#### 4.5 Discussion – Current Situation:

There was an awareness by the students of information being misplaced and the need for information to be repeated. The students were also aware of the size of the teams that were placed around them as well as questioning how effective the communication was between different organisations. Reference was made to the high level of communication that was required to keep this machine running. I sense there was a lot of guilt in the amount of resourcing that is required to ensure the flow is smooth of both information in and information out.

The interpretation of the parents is that they are often the support scaffolding that is built around an individual. They are the cog that ensures all the information is reaching the right destination in a timely manner. Along with the daily logistics of personal cares, transport arrangements and arranging support workers many hours are currently spent repeating formation that was passed on previously and liaising between the myriad of professionals who have been contracted to provide a service.

Support Workers often straddle the space between home and school. As mentioned earlier they are an integral part of the inner workings of support mechanism. In exploring the

current situation and some of the challenges that the support workers have the observations that have come to the fore are in information not being passed on and time given to planning.

The current situation that medical professionals experience within the context of this research is that information "is held in siloes and stands apart from that of other professionals that are working for the betterment of an individual" (Dorothy – Medical Professional). Medical Professionals are focused on the medical challenges an individual is experiencing and therefore leaves the social or holistic well-being up to those who work with the social services or education streams. One comment from one of the medical professionals stood above the rest when the data was being analyzed was "our system doesn't have a lot of good stuff". This statement aligns with the literature in claiming that the deficit model is still well and truly alive within the medical system (Shakespeare, 2013; Oliver, 2013).

The current situation theme of all five user groups shared an element of frustration. Although each group had different concerns, the overall sense was that either information was fragmented, or the way people advocated and engaged with each other was fragmented. The current situation across all groups identified that there is a huge amount of information that is lost or not passed on to the appropriate stakeholders. The number of people whom are supporting an individual is also large and the coordination of this can at times be challenging which has been highlighted. The current situation theme within the context of this research has been drawn on through the experiences and various situations that the interview participants are experiencing.

To delve deeper into the theme of *current situation*, I have looked into the research literature to attempt to understand what has brought us to where we find ourselves today in what is perceived as disjointed and difficult system to navigate (Selvaraj, 2015, Lyons, 2013, Ministry of Education, 2012). The area of research that holds the most relevance within this research is in the realm of the educational setting.

To avoid travelling too far away from my research question, I will concentrate briefly on two areas that do have significance.

1) What has brought us to our current situation?

#### 2) Inclusive Education – what does that even mean?

In discussing our current situation, we can look at the timeline that gives us an overview of the movement of special education or inclusive education as it is referred to in some circles.

By looking at the timeline and reading the work of Judith Selvaraj – *Inclusive education in New Zealand: rhetoric and reality* (Selvaraj, 2015), we can begin to understand that the current situation of working alongside people of all levels of ability in an attempt to be inclusive is not something that has happened, or not happened as some would argue, overnight.

In reading the history of the inclusive education movement and seeing what the government of the time was looking to implement we are able to gain a perspective that in striving towards a form of inclusion has been a work in progress for many years (Selvaraj, 2015).

"The Governments objective, broadly expressed, is that every person, whatever his level of academic ability, whether he be rich or poor, whether he live in the town or country, has a right as a citizen, to a free education of the kind for which he is best fitted and to the fullest of his powers ". (New Zealand Parliament, 1939, p. 59).

From this statement we can see that we have been working to understand this 'situation' for a long time.

Our current situation is the morphing of many years of policy shaping and reshaping in the hope that we arrive at a finish line called inclusive education. The complexities of our history within this space is not something that is easy to untangle or understand. It is a difficult space to traverse. It is taxing for those 'living' in this space to define where we are going when there is still so much debate around what we are trying to achieve.

In my segue from the current situation into the abyss of inclusive education the literature that accompanies this topic is growing exponentially. To once again keep my research tight, I have just touched on the million-dollar question of 'Inclusive Education – what does that even mean?' The research is not conclusive in that there is not a single statement that affirms this is what inclusive education is or that there is a single design on an inclusion model and what this looks like within the education setting. For the most part, people are generally on the same

playing field in agreeing to the concept but seeing the word on paper is where it stops. (Selvaraj, 2015, Lyons, 2013)

One premise that was noted and alluded to in Lesley Lyon's study – Transformed Understanding or Enlightened Ableism? The gap between policy and practice for children with disabilities in Aotearoa New Zealand, was the ideological versus reality of the situation of inclusive practice. Lyons (2013) stated,

They are required by legislation and policy to be inclusive of children with disabilities, and via this legislation and policy they have been given the language to speak about inclusion, yet making inclusion a reality appears problematic and confusing (p. 247).

In referring back to the findings of the current situation theme within this research project, the micro-level of a seemingly chaotic world for the participants and those whom are supporting them is reflected at a macro-level where there is disharmony and confusion in regards to the direction we should be heading.

#### 4.6 Advocacy

#### Student – Advocacy:

The enthusiasm shown by the students when discussing the advantages of having a site which would encourage personal advocacy was affirming. All the students recognised the sense of independence that could potentially be achieved if personalised information was housed on a site that they had access and control over. All three students were excited about the possibility by stating the "sense of freedom" that could be obtained as well as the "quality of life". A "better understanding of a person's needs" was also recognised as a benefit of self-advocacy. The students however didn't recognise advocacy as something beyond themselves. Their insights focused on what the platform would achieve for them as opposed to the world around them and those supporting them.

Person	Group	Quote
Huey	Student	"Manage our own profile"
Duey	Student	"Quality of Life/sense of freedom"
Luey	Student	"Better understanding of a person's needs"

## Parent – Advocacy:

The parents' observation were that communication around advocacy between themselves and their support networks was not ideal. The challenges in advocating for their sons and daughter were met with poor internal systems and admission of "information doesn't get shared with me having to be the medium all the time", "communication book, some people don't read it" as well as information across support organisations having their own unique way of recording information. An example of this was stated by one parent as having "four different places have different ways of recording information".

Person	Group	Quote
Fonzie	Parent	"Communication book, some people don't
		read it"
ChaiChi	Parent	"Information doesn't get shared without me
		having to be the medium all the time"
Joanie	Parent	"Four different places have different ways of
		recording information"

## Support Worker – Advocacy:

The advocacy theme among the support workers showed a variety of responses that alluded to the challenges of moving relevant information across networks. All groups recognised the trials of advocating on behalf of another person and went back to giving examples of the current situation. All groups made comments that could possibly fit under another theme of connection or ways to connect. The sharing of information and "emotional connections" was seen as a benefit of the concept as well as "no flow through". The advocacy theme among the professionals came from many angles.

Person	Group	Quote
Daisy	Support Worker	Emotional connections
Lily	Support Worker	Seeing what works
Violet	Support Worker	Anything that adds to communication

## **Teacher – Advocacy:**

The teachers' positioning on advocacy was that they understood the importance, however, were often stretched to be the child's advocate and to put in place measures where advocacy could be demonstrated at a high level. The teachers were able to give examples of when they tried to be positive advocates but the ability and time of others let the process down.

Person	Group	Quote
Harry	Teacher	Save emailing 8 people
Ron	Teacher	No flow through
Hermione	Teacher	Look back on history / what happened /
		what's been tried before

## Medical Professionals – Advocacy:

The medical professionals made reference to the system that they currently work within creates barriers for them to practice advocacy in such a way that they will see positive results. The duplication of information is a frustrating and the sharing of the same information time and time again appears to be frustrating also.

Person	Group	Quotes
Dorothy	Medical Professional	The current system is left behind
Rose	Medical Professional	What they do and how they do it gets lost
Blanch	Medical Professional	Telling story over and over
Sophia	Medical Professional	Having to reiterate information / not having to duplicate

## 4.7 Discussion - Advocacy

To understand the importance of advocacy and what it means to the individuals at the center of this website concept all participants made comments that there would be a positive outcome if such a concept was created. To be able to advocate accurately on a person's behalf is critical in truly hearing the voice of the individual. The interpretation of advocacy centered on high levels of frustration in what all groups were looking to achieve for the young people they were supporting. All highlighted various scenarios on how they were required to advocate and

the benefits of being a strong advocate. The challenges similar for all and pivoted around communication and how it was recorded or shared.

The purpose of exploring advocacy and the connection it has to this research is best described as something we are all doing continually for the people we work alongside. What has been identified in this research is that it is currently practiced in an *ad hoc* way with little knowledge of who is doing what, what has been done in the past, what is currently being done, who else or what other organisations are involved.

Regardless of your relationship with an individual we are all, for the most part, doing our best to advocate for the person with support needs to have a fulfilling and meaningful life. How we advocate and represent an individual who is unable to take on this role themselves is a huge honor which I think for many of us we can lose sight of in the business of our roles. For those professionals, whether that be support staff, teachers or medical professionals we are paid to give insight or pass on our opinion on how something ought to be managed for someone else. If you were to momentarily put yourself in the position of someone who was unable to speak or represent themselves would you like the person advocating for you to be informed, to know something about you or are you happy for them to decide on what they *think* is right for you based on 'feeling'. I believe most of us would opt for the first option if this scenario was presented.

The United Nations Convention article 23 emphasis the messaging of "nothing about us without us" (Moriarity & Dew, 2011). To be able to represent and advocate for people who are unable to advocate for themselves we must be able to capture their voices and wishes in such a way that they are as closely tied and unedited to the individual as possible. To build a rich tapestry and learn about the interests, traits and personality of an individual that requires additional support is an ongoing process that must work in unison with the individual (Protocol - Convention on the Rights of Persons with Disabilities and Optional Protocol UNITED *NATIONS*, n.d.)

The Social Care Institute for Excellence has identified that people with high and complex support needs "are likely to be left behind in social care provision" (Dalton & Hoyt-Hallett, 2013,

p. 32). Minimal resources are available, and the higher level of skill required to provide support is often absent. Self-advocacy among people with high and complex support needs, as supported by the investigation of Mithran (2009): "there is often a need for the involvement of an advocate to make their voices heard" (p. 32). The voices at the centre of this concept of a storing and sharing website are pivotal and at the forefront of our minds as we look to improve ways to engage and create meaningful experiences.

For many parents, the role of advocate becomes a full-time role, albeit a role that was not of their choosing. The success of their child's day is planned down to the *n*th degree with a large team supporting the smooth turning of the cogs. As Emily Pearl alluded to in her 'Welcome to Holland' poem, this journey wasn't the one that was booked however life must go on. There are windmills to see, tulips to pick, and Rembrandt, yes, even Rembrandts (Kingsley, n.d.). Being a parent comes with many challenges. Being a parent of a child with high and complex support needs comes with many additional challenges, as well as sleepless nights will only ever be understood if you are in that position yourself. The investment in time in raising a child regardless of ability is monumental. Achieving milestones, progressing alongside your peers within the often-unspoken perimeters of 'normal' is something that is taken for granted with limited reflection on what life might be like for those travelling on a different journey.

The advocacy required by parents to navigate the numerous systems they find themselves in commands a high level of both communication and negotiations skills. The resilience that is called upon daily to ensure the right messages are being disseminated can be, at times, overwhelming. The concept of a website whereby how and what we advocate as well as what is shared across our networks is something the parent group identified as having a positive influence. For advocacy to be powerful and on point for the people, it is intended for a streamlined approach would be beneficial for all. The parents spoken to are fully aware that they are their child's voice. They are aware fully aware of the responsibilities of being their child's advocate however are challenged daily around the logistics of doing this well. Chachi (parent P6) commented on "every year we lose information" and "have to reshare information all the time". This example shows that even though parents are doing their best to advocate for their children the system is disjointed and to some degree unreliable.

With the best of intentions parents are working hard to be the mechanism in keeping this machine running smoothly. To create a platform that becomes a *'one stop shop'* with information readily available would ensure an individual's support networks are kept updated and invited to view relevant pieces of information if appropriate. Joanie (parent, P10) stated that "one system would make life easier – heaps easier".

The position of a support worker can be described as a link between home and school although many other iterations are possible. The support workers role is one which is like that of the parent and can have a huge influence on the success of the day or any given moment within the day. Support Workers carry an enormous amount of information on the young person they are working alongside. Valuable information that is not called upon as often as it possibly could be for various reasons including the level of influence that is absent in their roles. Support Workers are the eyes, ears and voice and have an often-intrinsic understanding of what is going on for the young person that is built up over time. They are the key holders to what has worked well before and what does not work. The sharing of information across a team of people is critical for several support workers. Rose (Support Worker P3) reiterated the importance of being able to advocate stating "we need to know who what does", "it is vital if x (mum) is unable to be contacted we need to act in the moment".

Advocacy, within the concept of this research, requires information to flow. Unless there is the platform or infrastructure to support this the information will sit within a silo and the opportunity to bring a richness to an individual's life will potential be lost. To be able to share their learnings on a platform that enhances and celebrates the positive moments in a child's days would be of huge value. To be able to advocate and pass on information regardless of whether it is a small win, or a life changing moment would create more meaningful purpose for the support workers as well as a positive outcome for the young person they are advocating alongside. Rose (Support Worker P3) stated the importance of advocacy and how having the right information was important: "little things are important when you can't manage the big things" and "life is much richer when we see them as social beings not medical beings".

The role of advocate within the school setting often sits with the classroom teacher. Along with the thirty other students in the classroom, whom all require pastoral care to some degree, children with high and complex support needs require more. There can in some circumstances be a team of external providers who provide support in some form to the individual which is often delivered during the school day. The coordination of these services is done in collaboration with the classroom teacher.

For the teachers to advocate and bring meaning to a situation or an opportunity it is helpful if they have all the information available. This information might be on what has been tried before and worked or what has been tried previously and not worked. Gathering up this information can be time consuming of and this time is not something that teachers have a lot of. In trying to bring meaningful learning to a child, teachers work hard to find out what has happened in the past and build on that. This, however, isn't always easy. Advocacy can be thick with challenges within the school environment. A platform that captures who has done what, what has worked, what has not worked, who has been involved previously and from what service provider, would all be helpful information to have on hand. The building of this dossier of information over time would be beneficial for all parties.

The medical professionals working alongside an individual advocate with the medical model framework in mind (Shakespeare, 2013). The medical model, through the eyes of professionals who work in the social services, work to a deficit model and focuses on what is possibly wrong rather than looking at an individual holistically and all the things that are right or going well. In terms of advocacy and the researching of the storing and sharing concept there were some interesting comments shared by those working within the medical framework. Comments such as "our system doesn't have a lot of good stuff" Rose (Medical Professional P9) and Dorothy's (Medical Professional P8) comments of "limited ideas about what is possible" and "health information is degrading and restrictive about the future" captured the theme coming through from the participants from the medical professionals. Dorothy also made the comparison between teachers and teacher aides "teachers and teacher aides see possibilities whereas medical people tend to see in that very negative model – DIS ability model rather than

ABILITY model". The medical professionals were also aware of the repetition of information sharing.

Although the medical professionals were aware of their short comings in terms of holistic health and well-being, I don't believe their focus should or would change. What I do believe would be welcomed however is a snapshot of what else is happening in a person's world. Different decisions might be made when it comes to a person's medical wellness if this information was presented but without an overall view of what is happening in a person's life that insight and information will remain absent. As Sophia (Medical Professional) alluded to...." if only I'd known that". Oliver suggests that this could be seen as an opportunity to expand the viewpoint for the medical professionals and take into consideration the social model of disability (Ba, 1990).

Although not written with the context of New Zealand in mind Suzanna Rocco Dillion's (2013) paper, "'If not you, then who?' - Advocating for Physical Education Needs of Students with Disabilities'", makes a few points that are easily transferable in terms of advocacy and what this looks like for young people with disabilities. Dillon (2013) states that advocacy "takes commitment" which is clearly relatable through the findings of my research. Dillion (2013) also asks the question 'if not you, then who' (p. 39). Her question is pitched within the context of physical education; however, I would suggest this question could be used across all areas of an individual's life. Dillon (2013) cited in the National Association of Sport and Physical education (p. 11):

Who will protect the rights and interests of children within our physical education classes, and who will ensure *that all* children have an opportunity to be physically educated individuals with the knowledge, skills and confidence to enjoy a life time of healthful physical activity.

The relevance of my research reaffirms the importance of the papers title 'If not you, then who?'. Dillon also provides a list guiding principle on how to advocate effectively with examples of the sorts of policy and procedures individuals may find helpful. Although this paper has an American flavour to it the ideas would be easily transferable to a New Zealand context.
#### **Student / Parent – Empowerment:**

The messaging from both the student and the parent participant group was similar in that they understood the importance of empowerment and what such a notion would achieve for either themselves (students) or for their sons and daughters (parents). There were a range of responses across both groups on how sharing and storing relevant information on a website platform could be empowering for individuals. The parent group recognised the consequences of such and expressed the value of being able to "see their strengths and weaknesses" and that the "person that it's about is in control". The students used the word 'information' and "information sharing" a few times.

Both students and parents were able to make the connection that by sharing information a sense of empowerment would be attained.

Person	Group	Quotes
Huey	Student	So much to learn
Duey	Student	Access to important information
Luey	Student	Information sharing

Person	Group	Quotes
Fronzi	Parent	That the person that it's about is in control /
		it gives her a voice
Chachi	Parent	Record of the person / see their strengths
		and weaknesses
Joanie	Parent	Knowing background / personal approach

#### Support Worker/ Teacher / Medical Professional – Empowerment:

The support workers were able to make the connection between the concept of the site and how this would bring a sense of empowerment. The support workers suggested that going forward this is going to be very beneficial.

Person	Group	Quote
Daisy	Support Worker	What provides quality of life
Lily	Support Worker	Get ideas on how to implement strategies
Violet	Support Worker	Going forward for the child this is going to be very beneficial

#### **Teacher – Empowerment:**

Both teachers were able to give examples of what actions would be required to achieve a sense of empowerment for the individuals they support. Being able to share examples of what provides quality of life along with how to get ideas on how to implement strategies

Person	Group	Quotes
Harry	Teacher	Moments to show off / accessing ideas
Hermione	Teacher	Record of kids portfolio

#### **Medical Professional – Empowerment:**

The findings in relation to empowerment within the medical professionals was limited. There were positive examples of how empowerment would be achieved using the platform and examples given were in how engaging with parents more readily would create a sense of empowerment.

Person	Group	Quotes
Rose	Medical Professional	"Parents in control which is a good thing".

#### 4.9 Discussion - Empowerment

We can't underestimate the sense and impact of empowerment. To be able to have the information on-hand and pick up where someone else has left off is hugely empowering. To also discover what works, how it works and capture that for others to benefit from is not only empowering but also extremely rewarding.

The parent group were vocal in how they viewed the concept and how this would lead to empowering their children. All parents were positive in that it was the child at the centre of the concept website and that they would be in control of who sees what information. The overarching view was that the "more information the better" and that people were "crying out for information" Chachi (parent P6). Being in a position where they were able to witness the trials and tribulations of information sharing and the benefits this would bring to their children the concept was well supported.

The support workers view in relation to how empowering opportunities would be created if personalized information was on hand was described best by Lily saying that "if you can get one idea out of it it's one idea you didn't have before".

Like advocacy responsibilities, we are consistently looking for prospects where we can create present, participating and learning opportunities which in turn is empowering.

Lily (Support Worker P7) described the benefits of the concept and how it could be used to empower an individual as a "one stop shop".

Empowerment opportunities must be embraced when presented within this sector and particularly within this group of young people. The empowerment opportunities are not glaringly obvious therefore when identified need to be nurtured. For children with high and complex support needs 90 percent, maybe higher, of what they do each day is decided for them by others. From the moment they wake up someone else will decide what they wear, what they will eat for breakfast, when they will go to the bathroom, how they'll travel to school and what they will do once they get there. All of this is before the 9 o'clock bell goes.

Empowerment can be subtle, and communication can be subtle, but we must look for it. For a child that is unable to vocalize their wishes and uses a nod, a blink or the poke of a tongue to convey a message, tuning into this unspoken form of communication can in turn lead to the opportunity to empower. To decipher that the blink of an eye, the twitch of the head or the poke of a tongue all mean 'yes' we can begin to build a rich tapestry of ideas for others to use and continue to build a dossier of tailored knowledge. To have accurate information on hand is hugely empowering for the individual. As a professional you are saying or demonstrating that "I know you" and "I know what you are capable of". As opposed to "now where did we get to last term" or "what did your last school do".

75

# 4.10 Collaboration

#### **Student – Collaboration:**

The student group didn't use the terminology of collaboration *per se;* however, one student was able to make the connection that the website would allow for information to be distributed and "education" across their support network.

Person	Group	Quotes
Luey	Student	Education

# **Parent – Collaboration:**

One parent talked about the continuity among professionals as being a positive outcome in using the platform as well as the platform being a place that everyone can access.

Person	Group	Quotes
Fonzie	Parent	Continuity among professionals / everyone
		can access

#### **Teacher – Collaboration:**

The teachers identified several practical advantages in using the platform. Examples of these are "saves emailing eight people", "communication out could be really successful" as well as stating that it would "get those professionals talking to each other as required or needed".

Person	Group	Quotes
Harry	Teacher	Save emailing eight people / notifications / accessibility of it is huge / communication out could be really successful / communication in huge teams
Hermione	Teacher	I think it is quite good because it gets those professionals talking to each other as required or needed.
Ron	Teacher	Transition

## **Support Worker – Collaboration:**

The support workers identified that the benefits would be being able to draw on the different expertise of people right across the individuals support network. It could potentially act as a "one stop shop".

Person	Group	Quotes
Lily	Support Worker	Information isn't passed on / one stop shop.

## **Medical Professional – Collaboration:**

One medical professional confessed that of the "4 to 5 different professionals and none of them meet" as well as a second medical professional sharing that "what they do and how they do it gets lost".

Person	Group	Quotes
Rose	Medical Professional	What they do and how they do it gets lost.
Blanche	Medical Professional	4/5 different professionals and none of them meet.

#### 4.11 Discussion - Collaboration

The teachers were able to see the benefits of having the storing and sharing website and how this would be of benefit not only within the education system but the more comprehensive network that engage with an individual.

One teacher made the connection that in having access to a website that held information on an individual and that the information was made available to different support networks when appropriate would be of benefit, "I think it is quite good because it gets those professionals talking to each other as required or needed". This was backed up by one of the medical professionals with her statement of "4 - 5 different professionals and none of them meet".

Another teacher was excited by the thought that his days of emailing eight different people might be coming to an end and that the site would also have the ability to notify him if relevant information was uploaded. Harry recognised that not only would the site have information coming in, it would also be a means of getting information out to relevant parties. The benefits of communicating with huge teams was also alluded to. All participant groups understood and saw the benefits of working collaboratively.

With the best of intentions most individuals working alongside and advocating for people with high and complex support needs are open to and do see the benefits of collaboration. However, the reality of a collaborative approach varies greatly among the different support networks involved in a person with high and complex support needs life.

The benefits of working collaboratively within the mainstream population is huge and great things are achieved when stakeholders come together for the greater good of a cause or movement. To work towards some form of high-level collaboration is, in my view, more so important when the person at the centre is unable to present or represent their views and thoughts independently.

In order to work collaboratively and share information, we need to have access to information in the first instance. The current situation which has been referred to throughout this research is that it is "pear shaped, and everything is bits of paper with no flow on". It can be extremely challenging to pass on information when in fact there is nothing recorded, or people have moved on. Building a dossier of information for people with high and complex support needs is crucial in building what can be termed as a "good life". A rich file of information that can be created over a person's life and then being able to share this information where appropriate allows a person's support network to see the person as an individual who has had experiences that can be built on as opposed to starting from scratch at each turn.

A large body of literature has been researched with the theme of *community conversations* emphasizing the importance of across organisational collaborative engagement. In the Journal of Disability Policy Studies, Audrey Trainor (2018) describes community conversations "by harnessing the expertise and motivation of key stakeholders" (p. 2). This concept of 'community conversations' would support the concept of my research topic in that it encourages the sharing of ideas across an individual support network.

78

Although the concept of Community Conversation is met with the best of intentions Carter (2018) emphasized that this model of practice may be ineffective if delivered in insolation. The Community Conversation model and the collaboration of stakeholders is effective at "spurring change" (p. 199) however the sustainability of such a model is currently unknown.

Supporting the Community Conversation research theme and collaborative partnership model, the study by Flippo and Butterworth, (2018) shared a similar belief of Carter (2018), in that the concept generated by the community conversation model was effective as a "catalyst for system change" (p. 10). Flippo and Butterworth (2018) lay emphasis on the importance of a holistic approach across all stakeholders is required for change to take place.

#### 4.12 Resources

#### **Teacher and Parent - Resources:**

Only the teachers were able to identify a site that is used within the educational space and this site is used across all students or all abilities. This site is called Seesaw. There was no other reference to a storing and sharing sites across any of the other participant groups. One parent referred to this with a statement of "I don't think there is anything else out there" during R Q 1.

Person	Group	Quotes
Harry	Teacher	Like seesaw but extends to a
		border people.
Hermione	Teacher	Seesaw
Fonzie	Parent	I don't think there is anything
		else out there.

#### 4.13 Discussion - Resources

The resource theme did not emerge with a lot of information which was enlightening and may have been a good reason to leave it out as a theme. However, the deficiency of knowledge gave more reason to include 'resource' as a theme. In speaking with all participant groups and for there to be such minimal reference to a platform where information can be shared and stored was helpful. Across the six interview questions there was very limited information coming forward on how information is shared and stored for our most vulnerable students. The effort and energy that is required to manage the logistic of the day is what the focus appears to be on as opposed to researching what assistive technology or any other resources might be available is not a priority. This information showed that there potentially is an opportunity to take this concept to the next phase.

Upon reflection, this could or should have been a direct research question. It may have triggered the memory of the participants and we could have drilled down more into what they knew rather than it floating to the surface nonchalantly and then being recognised as an important piece of information.

Unless there is a high level of advocacy from parents, caregivers or support networks who have the skills to be proactive in terms of seeking out information and tailoring it to the needs of the individual then seems to be a *laisse faire* approach.

The awareness of what is available for individual and families across the assistive technology environment is minimal.

Assistive Technology for people with high and complex support needs sits at what could be described at the pointy end of the stick. The population of users is a lot smaller and the intricacies of specific designs is high end and tailored to the individual as opposed to off the shelf software or an off the shelf product.

The absence of ideas across all participant groups on what resources are available showed evidence that there is in fact nothing available on the market or that there potentially are products available, but the support networks are not able to seek out what is available. 4.14 Accessibility

#### **Student – Accessibility:**

All the student participants spoke from a personal point of view in sharing what they would or do find helpful in accessing various websites. The students covered all the traditional accessibility features including larger font, eye tracking or eye gaze as well as both voice activation and facial recognition.

Person	Group	Quotes
Huey	Student	Large font / voice activation
Duey	Student	Facial recognition / eye tracking / audio cues / quick search / universal design.
Luey	Student	Eye gaze

#### **Parent – Accessibility:**

The parent group was particularly insightful and had clearly experienced assistive technology in all forms. Feedback included the size of the buttons; a clean and simple design was preferred over something busy. Compatibility across various devices including it being accessible through an app was emphasized.

Person	Group	Quotes
Fonzie	Parent	Big buttons / functional on iPad / not too much noise / buttons not too close together / clean / simple.
Chaichi	Parent	Make it so it's an app / touch screen rather than mouse.
Joanie	Parent	Needs to be accessed by every form of communication device / compatible with all different forms of software.

#### **Teacher – Accessibility:**

The teachers alluded to the platform being accessible from a "not being blocked on the server" angle. They all gave examples of what accessibility looked like from an end user view point however the main focus was on the "ease of access", "easy to upload" and "available to everyone" were the key messages.

Person	Group	Quotes
Harry	Teacher	Font / hearing considerations / some form of translate.
Ron	Teacher	Not blocked on server / easy to upload / don't create work for teachers / available to everyone.
Hermione	Teacher	Visual aids / ease of access.

#### **Support Worker – Accessibility:**

Like both the student and the parents the support workers mentioned the traditional forms of assistive technology necessities including "very succinct clear and concise" layout as well as adding the idea of "ways to personalize it" and "take ownership of it".

Person	Group	Quotes
Daisy	Support Worker	Ways to personalize it / take ownership of it.
Lily	Support Worker	Larger font.
Violet	Support Worker	Keep English very simple / very succinct, clear and concise.

## Medical Professional – Accessibility:

Three out of the four medical professionals indicated the need for the platform to be "very simple" with "basic instructions". The idea of "pictorial clips of young people doing things" or "something that shows up their strengths" was also discussed. Like the other participant groups ideas such as "visual, but not too visual", "eye gaze" and "voice activation" was raised.

Person	Group	Quotes
Dorothy	Medical Prof	Graphs, tables. Pictorial, video clips of young people doing things / not long written reports / introductory page / something that shows up strengths.
Rose	Medical Prof	Visual without being over visual / sensory issues / if too complicated we won't use it.
Blanch	Medical Prof	Vision / voice activated instructions / basic instructions.
Sophia	Medical Prof	Eye gaze / font size / very simple.

# 4.15 Discussion - Accessibility

The responses from all interview participants on the importance of accessibility of the platform and the types of features that needed to be considered was highly populated. The

suggestions ranged from font size and colour through to eye-gaze options. There were many suggestions with each user group having a specific interest depending on their own personal or professional experience.

The findings across all participants covered a range of topics and point of interest. Each group was coming from their own lived experience and what mattered to them as either an individual with additional support requirements or as a person who supports provides support.

All groups had an awareness of the traditional forms of assistive technology such as voice activation and eye gaze. The parental group had a much higher awareness of what would better serve their young people such as the larger buttons, less chaotic screens and device compatibility. The accessibility of assistive technology varies greatly across individuals with additional support needs.

The term 'assistive technology' was first mooted in 1988 in the Technology Related Assistance for Individuals with Disabilities Act. It was amended and replaced in 1998 with the Assistive Technology Act 1998 and has been described as "any item, piece of equipment, or product system, whether acquired commercially, modified or customized that is used to increase, maintain or improve functionality capabilities of individuals with disabilities" (p. 26) (Assistive Technology Act, 1998 cited in Assistive Technology – Fixing Humans) (Katsioloudis & Jones, 2013).

The various options of accessibility features have been discussed by Katsioloudis and Jones (2013) with examples given across a spectrum of disabilities. Examples include computerassistive technology where eye gaze and breath are features, through to augmentative and alternative communication devices (AAC). AAC is using voice to text or text to voice depending on the requirements of the individual (Katsioloudis & Jones, 2013).

The functionality of a platform particularly for a person with high and complex needs is made or stalled in this initial stage "is it accessible to me or not"? The design around the personalized accessibility of the platform needs to have many considerations made to it.

Due to the sometimes-complicated nature of an individual with high and complex needs the accessible considerations fall outside the generic programming of someone with less support needs? The recommendations made by the participants alluded to accessible ideas that fall outside the 'off the shelf' models. Eye gaze and voice activation would be examples of this. The tailored needs must be taken into consideration for the platform to be accessible. Without this the site is redundant for the individual.

Consideration must be well thought out regarding the layout of the platform to ensure easy navigation for the end user. The opportunity to create a personalized platform would give a strong sense of empowerment to the end user.

The opportunity to explore the current situation and seek out other platforms that are being used by individuals with high and complex support needs within the assistive technology world is exciting. In relation to assistive technology Paul Wise's research (Wise, 2010), *Emerging Technologies and their impact of disability,* discusses the influence assistive technology has on children with disabilities and the equality that is being leveraged through the advancement of assistive technology. Wise (2010) states that "all technologies attempt to address some deficiency in human capacity or the human condition" (p. 171). Wise also identifies the "rapid evolution" of assistive technology that is currently being produced at an expediential rate. Another claim that Wise (2010) makes is in the use of his term 'social engagement' and the benefits of assistive technology has for all.

#### 4.16 Security

#### Student – Security:

The students were very aware of the risks of having such sensitive information stored on a cloud base platform. All the students were able to articulate examples of security risks and what these might present as. Examples that were shared in relation to security were "possible misuse of information", "wrong people gaining access", "storing sensitive information with certain people".

Person	Group	Quotes
Huey	Student	Access to information / information getting deleted.
Duey	Student	Possible misuse of information / wrong people gaining access / getting people to store in one location could be an issue / storing sensitive information with certain people.
Luey	Student	Personal information remains private / privacy.

#### Parents – Security:

Both parents identified privacy as being a security concern or other people accessing their child's information. The suggestion of having all people who become a part of a young person's platform sign a waiver was suggested.

Person	Group	Quotes
Chaichi	Parent	Others accessing information.
Joanie	Parent	Privacy issues / sign a waiver.

#### **Teachers – Security:**

The teachers shared the same concern as the parents regarding how safe the information would be and what restrictions would be in place to ensure the information housed on the site was secure. "Safety of information" was expressed.

Person	Group	Quotes
Harry	Teacher	Safety of information / restrictions were adhered to.

# **Support Worker – Security:**

The theme of permission or "giving permission" was indicated as a concern by a support worker as well as having to negotiate through "red tape" to have access to the platform. The reference to red tape is in relation people accessing the site and the permission settings.

Person	Group	Quotes
Daisy	Support Worker	Personal intent / red tape.
Lily	Support Worker	Giving permission

#### **Medical Professional – Security:**

The same concerns were shared among the medical professionals regarding confidentiality as well as the additional concern of "hackers".

Person	Group	Quotes
Dorothy	Medical Prof	
Rose	Medical Prof	Safe from hackers / have good profiles to show how to use it / privacy.
Blanche	Medical Prof	Confidential / privacy and confidential.
Sophia	Medical Prof	Security.

#### 4.17 Discussion - Security

Security was a concern right across all groups with a range of examples given. There was nothing hugely surprising coming out of the theme around security. All participants shared concerns around the level of privacy the individuals would have in using a concept such as the one being proposed and how this would be managed. Some valuable insights were gained in speaking particularly to the young people themselves and how they viewed their own information and the importance of both privacy and security.

The protection of personal information is something very much at the forefront of our minds for most individuals when we are housing information online. So much of what we do and who we are in housed in a cloud base website. There will always be an element of risk around this.

The security of websites and how this is managed is a separate piece of research however as stated throughout the literature the benefits of housing information and sharing it across a cloud based platform, far outweigh the risk (Petroulakis, 2017; Wilson, 2011). Understanding there is a concern is important and that there are ways for this to be managed. Exploring how this could potentially be administered falls outside this immediate research due to the enormity of it. This research is testing the concept of an idea the logistics are not ready to be investigated at present. The feedback from all participant groups indicated some form of security being of concern to them.

# **Student – Barriers:**

The students were aware of the barriers that may be presented in getting people to agree to use the site and stated that some medical people may "prefer the traditional method of paper". The accuracy of information was also stated as a potential obstacle.

Person	Group	Quotes
Huey	Student	Getting professionals such as medical people
		to agree to use it as some might prefer the
		traditional paper method.
Duey	Student	Making sure information is accurate.

## **Parent – Barriers:**

Each of the parents had different views on what the possible barriers could arise ranging from "people who decide it wouldn't work because it's never been used that way" through to the "marketing" of the site and "letting people know how it works". The "funding" of the site was also raised as an obstacle.

Person	Group	Quotes
Fonzie	Parent	Maintaining passion and commitment to it / get people who own it to help drive it / marketing and letting people know it works.
Chaichi	Parent	Funding it / people who don't want others to have access to their information.
Joanie	Parent	People who decide it wouldn't work because it's never been used that way.

#### **Teacher – Barriers:**

The teacher raised concern around the "buy in" from people and getting people to use the site. "Red tape" was also mentioned in the context of people having easy access to the site. One teacher raised the question around ownership of the site and maintained it "can't become owned by one of these single entities".

Person	Group	Quotes
Harry	Teacher	Buy in from people / it can't become owned
		by one of these single entities.
Harry	Teacher	Funding it / buy in / red tape.
Harry	Teacher	Buy in / Time.

#### **Support Worker – Barriers:**

Like the teachers the support workers identified three different areas that could create possible obstacles. The "amount of different people doing it" was seen as an obstacle as well at the platform being "really complex". The "time" it takes to do and "people see it as another job" was also raised as an obstacle.

Person	Group	Quotes		
Daisy	Support Worker	Complexities / the amount of different people doing it / really complex		
Lily	Support Worker	Buy in / time / paperwork.		
Violet	Support Worker	People see it as another job to do / attract IT difficulties /		

## **Medical Professional – Barriers:**

One medical professional declared that the Barriers will be within their own sector and that the medical sector are very "wary of change" with the challenge of convincing "them (medical professionals) that this is a good idea". Having the opportunity to show that this is going to be useful" and that "once you get it up and running it will save time in the long run" was also shared as a possible solution to the obstacle.

Person	Group	Quotes
Dorothy	Medical Prof	People are very busy / stay as clear as you presented it / logical / prompt to say what is available.
Rose	Medical Prof	To show how it can be used.
Blanch	Medical Prof	Challenge will come with professionals / wary of change / convince them this is a good idea / best person to sell it will be the person themselves.
Sophia	Medical Prof	Showing that it is going to be useful / everyone has to buy in / once you have it up and running it will save you time in the long run.

# 4.19 Discussion - Barriers

There were numerous barriers mentioned across all groups however the main area of concern was getting buy in from the relevant parties. Each group alluded to this in different guises, but the messaging was similar. Getting the buy in from an already busy sector was going to present an obstacle. Several obstacles spoken to were also aligned with a potential solution

such as "it will save time in the long run" and "the best person to sell it will be the person themselves".

There were also some valid points in remaining passionate for the concept as such a task can take many years to complete as well as bringing people around to the idea of using something different.

The barriers that were presented across all user groups did not seem insurmountable. If the product was built with the recommendations made by the participants groups, then several the obstacles would resolve themselves. The suggestion by one parent of "get people who own it to help drive it" would help alleviate several of the obstacles. For people to understand the importance of this information sharing and storing site and the benefits it will bring to the individual at the center are best explained by the individual themselves.

#### 4.20 Summary of Findings

In summarising the findings of this research, there is clearly not only a concern for the current situation but also a willingness to look for a different approach. Although there were five different participant groups the overall messaging was that things could improve. Each group had different areas that were important or relevant to them however the overall messaging was similar in that the across sector the storing and sharing of information using a collaborative approach is lacking.

Throughout the findings reference is made numerous times to the benefits of being able to share and store information and how this would improve the lives of people with high and complex support needs. The sense of empowerment is also highlighted across all participant groups and seen as something that would work in favor of the individual. The absence of any references by the participants to a resource that would encourage the sharing and storing of information did not go unnoticed. The void in mentioning such a platform was noted across all participant groups. Accessibility and the user-friendliness of a potential site gained many suggestions on how this could be achieved. Simplicity of both imagery and accessibility have come through as key considerations.

Security of personal information was emphasised particularly from both the individuals and the parents. The main area of concern was in the personal information falling into the wrong hands which is a significant concern. Security was also discussed through the other user groups although this was in relation to having the ability to share information when government systems are so locked down. This finding has been valuable and will now help in giving guidance in how information is shared, if the idea is developed further, across support organisations.

Organisational buy-in featured as a barrier in moving this concept to a phase where both individuals and organisations engaged with it. Changing the way systems are used is difficult and this was acknowledged. Unless a concept reduces workload rather than increases work load it would be hard to convince parties to use it.

# **5 CHAPTER FIVE:**

## 5.1 Evaluation of Study

In gaining an appreciation of the complexities across all spheres of the research I can see that there is room for improvement in the way we share information and how decisions are made for or on behalf of an individual. There are clear benefits in having a high functioning collaborative approach across a person's support network. In looking at both the tangible and intangible threads within this research there is a commonality in that all parties are looking to find meaningful ways to support people with high and complex needs to the best of their ability. The technology industry is making positive gains in high end individually tailored assistive technology that can be controlled through a variety of methods. Stakeholders are working in an environment where expectations are adjusted to fit the fiscal constraints while still bringing hope to a situation. A solution focus approach has been adopted.

The concept of a storing and sharing platform is sound. The research findings and supporting evidence have identified there is a place for such a platform. However, it would be naïve to think that such an idea would be easy to implement. The strains on an already saturated care and education industry are stretched. For such a concept to be implemented several factors would need to be considered. These factors have come from the research findings and include the simplicity of the platform as a key consideration as well as the security of personal information. The current concept I believe is too complicated. As one person has said to me who wasn't part of the research group "I'd need to be able to use it while standing in line at the supermarket". I do not believe the current concept is pitched at this level.

I believe this research has served the purpose of testing the concept of a storing and sharing platform as well as assisting in giving context to the larger question of how we have arrived at our current destination. In understanding the context and knowing that there is systemic confusion in the inclusive education space brings the challenge of building collaborative conversations, regardless of the method, into perspective. I believe this research has shown that there is the opportunity to create a resource that information could possibly be stored. Do I believe this concept will bring clarity to inclusive practice on a nationwide scale – no I don't! That was never the intention. Has this research help build a picture of what I am up against – yes it has! Do I believe such a resource could help a small portion of the population share information that is relevant to them having a meaningful opportunity – yes I do!

In order to take this concept to the next phase more research needs to be carried out with a smaller targeted group with less questions. The research question I presented had two parts to it which I think blurred the research for me as a new researcher. I was focused for a long time on the 'who' would benefit from the concept as opposed to the actual concept itself. I don't believe anything has been lost apart from my own time, however a huge amount has been gained in terms of my own learnings and the importance of keeping things extremely focused.

## 5.2 Recommendations

With the evidence gained through my research project, my recommendation is to continue into a development phase by building a prototype that could be tested in a controlled type situation. The original concept would be required to be pegged back somewhat to a simpler version with the overriding thought and example given that individuals and their advocates need to be able to use it while standing in line at the checkout. This is how simple it needs to be.

There are going to be some financial constraints on moving the concept to the next phase which needs to be explored. There are several startups programmes that could be investigated. There is also a company in Dunedin that may consider the project *pro bono*.

Ideally, an end user group needs to be formed to give guidance to the project. This was recommended in the findings.

In terms of the functionality of the concept my recommendation is that all information that is potentially shared is put through an authorization type process where the owner of the site can check the information being shared before it is made live. This recommendation comes after considering the feedback from the participants. In suggesting this it would also create a more streamline way of seeking permission and remove the hoops that would be fought if an organisation tried to share information.

A final recommendation would be to share these research findings with the Ministry of Education – Learning Support, Inclusive Education Action Group and the Complex Carers' Network which may generate some thoughts and discussions on how things could be done differently. From the research findings the evidence shows that the current situation is not meeting the needs of our most vulnerable students. Learning Support has received a significant increase in funding for students with additional support needs, therefore there may be an opportunity to leverage this concept further. The Well-Being budget (Labour, 2019) may also support an initiative to trial a pilot programme whereby the concept could also be tested.

# 6 **REFERENCES**:

- Article 1 Purpose | United Nations Enable. (n.d.). Retrieved June 4, 2019, from https://www.un.org/development/desa/disabilities/convention-on-the-rights-of-personswith-disabilities/article-1-purpose.html
- Article 21 Freedom of expression and opinion, and access to information | United Nations Enable. (n.d.). Retrieved June 4, 2019, from https://www.un.org/development/desa/disabilities/convention-on-the-rights-of-personswith-disabilities/article-21-freedom-of-expression-and-opinion-and-access-toinformation.html
- Article 24 Education | United Nations Enable. (n.d.). Retrieved June 4, 2019, from https://www.un.org/development/desa/disabilities/convention-on-the-rights-of-personswith-disabilities/article-24-education.html
- Article 30 Participation in cultural life, recreation, leisure and sport | United Nations Enable.
   (n.d.). Retrieved June 4, 2019, from
   https://www.un.org/development/desa/disabilities/convention-on-the-rights-of-persons-

with-disabilities/article-30-participation-in-cultural-life-recreation-leisure-and-sport.html

- Azhar, S., Ahmad, I., & Sein, M. K. (2010). Action Research as a Proactive Research Method for Construction Engineering and Management. *Journal of Construction Engineering and Management*, 136(1), 87–98. https://doi.org/10.1061/(ASCE)CO.1943-7862.0000081
- Ba, M. O. (1990). *The individual and social models of disability*. Retrieved from https://disabilitystudies.leeds.ac.uk/wp-content/uploads/sites/40/library/Oliver-in-soc-dis.pdf
- Batorowicz, B., Mcdougall, S., & Shepherd, T. A. (2006). AAC and community partnerships: The participation path to community inclusion. *Augmentative and Alternative Communication*, *22*(3), 178–195. https://doi.org/10.1080/07434610500468498
- Beyer, S., & Perry, J. (2013). Promoting independence through the use of assistive technology.
   *Tizard Learning Disability Review*, *18*(4), 179–185. https://doi.org/10.1108/TLDR-03-2013-0028
- Bradbury, H. (n.d.). The SAGE handbook of action research.
- BuketYükselAlptekinKüpçüÖznurÖzkasap. (2017). Research issues for privacy and security of electronic health services. *Elsevier*, 68, 1–13.
- Carter, E. W., & Bumble, J. L. (2018). The Promise and Possibilities of Community Conversations:
   Expanding Opportunities for People With Disabilities. *Journal of Disability Policy Studies*, 28(4), 195–202. https://doi.org/10.1177/1044207317739408
- Ching Hsiang Shih. (n.d.). Assisting people with disability improve collaborative pointing with mouse scroll wheel.
- Coleman, E. A. (2003). Falling Through the Cracks: Challenges and Opportunities for Improving Transitional Care for Persons with Continuous Complex Care Needs. *Journal of the American Geriatrics Society*, *51*(4), 549–555. https://doi.org/10.1046/j.1532-5415.2003.51185.x
- Coles, J. (2001). The Social Model of Disability: What does it mean for practice in services for people with learning difficulties? The Social Model of Disability: what does it mean for

practice in services for people with learning diff culties? *Disability & Society*, *16*(4), 501–510. https://doi.org/10.1080/09687590120059504

- Comparisons between the medical and social models of disability discourse. | Download Table. (n.d.). Retrieved July 1, 2019, from https://www.researchgate.net/figure/Comparisonsbetween-the-medical-and-social-models-of-disability-discourse\_tbl1\_297722476
- Convention on the Rights of Persons with Disabilities Articles | United Nations Enable. (n.d.). Retrieved June 4, 2019, from https://www.un.org/development/desa/disabilities/convention-on-the-rights-of-personswith-disabilities/convention-on-the-rights-of-persons-with-disabilities-2.html
- Dalton, C., & Hoyt-Hallett, G. (2013). Enablement through Provision of Assistive Technology: Case Reports of Two Children with Physical Disabilities. *British Journal of Occupational Therapy*, *76*(2), 108–111. https://doi.org/10.4276/030802213X13603244419356
- Dillon, S. R. (2013). "If Not You, Then Who?" Advocating for the Physical Education Needs of Students with Disabilities. *Strategies*, *26*(4), 38–39. https://doi.org/10.1080/08924562.2013.799933
- Fenwick, T. J. (2000). Expanding Conceptions of Experiential Learning: A Review of the Five Contemporary Perspectives on Cognition. *Adult Education Quarterly*, *50*(4), 243–272. https://doi.org/10.1177/07417130022087035
- Flippo, K., & Butterworth, J. (2018). Community Conversations and Transition Systems Change. Journal of Disability Policy Studies, 29(1), 7–11. https://doi.org/10.1177/1044207317739404
- *Foreword 2 Introduction 3*. (n.d.). Retrieved from http://www.hcn.govt.nz/documents/the-hcnunit/high-and-complex-needs-interagency-strategy.pdf
- Fundamental Principles of Disability National Disability Arts Collection & amp; Archive. (n.d.). Retrieved June 6, 2019, from https://the-ndaca.org/resources/audio-describedgallery/fundamental-principles-of-disability/

Golom, Frank, S. J. S. (2018). The Journey to Interprofessional Collaborative Practice-ClinicalKey. Retrieved June 7, 2019, from https://www-clinicalkey-comau.ezproxy.otago.ac.nz/#!/content/playContent/1-s2.0-S0031395517301335?returnurl=null&referrer=null

Hammel, J. (n.d.). *Technology and the environment*.

- Heah, T., Case, T., McGuire, B., & Law, M. (2006). Successful Participation: The Lived Experience among Children with Disabilities. *Canadian Journal of Occupational Therapy*, 73(4\_suppl), 1–10. https://doi.org/10.2182/cjot.06.010
- Healey-Ogden, M. J., & Austin, W. J. (n.d.). Uncovering the Lived Experience of Well-Being. *Qualitative Health Research*, *21*(1), 85–96. https://doi.org/10.1177/1049732310379113
- Heery, E., & Noon, M. (2017a). A dictionary of human resource management. Oxford University Press.
- Heery, E., & Noon, M. (2017b). A Dictionary of Human Resource Management (Vol. 1). https://doi.org/10.1093/acref/9780191827822.001.0001
- History Of Inclusive Education In New Zealand Timeline | Preceden. (n.d.). Retrieved June 4, 2019, from https://www.preceden.com/timelines/134431-history-of-inclusive-education-in-new-zealand
- History of the UN | United Nations Seventieth Anniversary. (n.d.). Retrieved June 4, 2019, from https://www.un.org/un70/en/content/history/index.html
- Katsioloudis, & Jones, M. (2013). Assistive Technology Fixing Humans. *Resources in Technology and Engineering*.
- Kingsley, E. P. (n.d.). *Texas Parent to Parent Welcome to Holland*. Retrieved from www.txp2p.org
- Lancioni, G. E., O, M. F., Singh, N. N., Sigafoos, J., Boccasini, A., La Martire, M. L., ... D, L. F.
  (2016). Technology to support positive occupational engagement and communication in persons with multiple disabilities. *Int J Disabil Hum Dev*, *15*(1), 111–116.

https://doi.org/10.1515/ijdhd-2015-0023

- Lundgren, H., Bang, A., Justice, S. B., Marsick, V. J., Poell, R. F., Yorks, L., ... Sung, S. (2017). Conceptualizing reflection in experience-based workplace learning. *Human Resource Development International*, 1–22. https://doi.org/10.1080/13678868.2017.1308717
- Lyons, L. (2013). Transformed Understanding or Enlightened Ableism? The Gap Between Policy and Practice for Children with Disabilities in Aotearoa New Zealand. *International Journal of Early Childhood*, 45(2), 237–249. https://doi.org/10.1007/s13158-013-0086-1

MandiaMentisWendyHolley-

BoenPhilippaButlerAlisonKearneyJuliaBuddTracyRileyJudeMacArthurVijayaDharanJillBevan -Brown. (2016). *mawhai: Webbing a professional identity through networkd interprofessional communities of practice*.

- McNiff, J., Edvardsen, O., & Steinholt, M. (2018). 'Impact', educational influence and the practice of shared expertise. *Educational Action Research*, 26(5), 803–819. https://doi.org/10.1080/09650792.2018.1426469
- Medicine, Social, Green, transparent png image & amp; clipart free download. (n.d.). Retrieved June 7, 2019, from https://www.kissclipart.com/medical-model-social-model-clipartsocial-model-of-ao7nz3/
- Ministry of Education, M. (2012). *Ministry of Education Supports and Services for Learners with Special Education Needs/Disabilities*. Retrieved from www.educationcounts.govt.nz
- Mithran, S. (2009). Self-advocacy for people with high support needs ProQuest. Retrieved May 27, 2019, from https://search-proquestcom.ezproxy.otago.ac.nz/docview/223289329?rfr id=info%3Axri%2Fsid%3Aprimo
- Miura, T., Yabu, K., Ikematsu, S., Kano, A., Ueda, M., Suzuki, J., ... Ifukube, T. (2012). Barrier-free walk: A social sharing platform of barrier-free information for sensory/physically-impaired and aged people. 2012 IEEE International Conference on Systems, Man, and Cybernetics (SMC), 2927–2932. https://doi.org/10.1109/ICSMC.2012.6378238

- Molyneux, J. (2001). Interprofessional teamworking: what makes teams work well? *Journal of Interprofessional Care*, 15(1), 29–35. https://doi.org/10.1080/13561820020022855
- Moriarity, L., & Dew, K. (2011). The United Nations Convention on the Rights of Persons with Disabilities and participation in Aotearoa New Zealand. *Disability & Society*, *26*(6), 683– 697. https://doi.org/10.1080/09687599.2011.602861
- New Zealand. Statistics New Zealand. (n.d.). *Disability survey, 2013.* Retrieved from http://archive.stats.govt.nz/browse\_for\_stats/health/disabilities/DisabilitySurvey\_HOTP20 13.aspx
- New Zealand Disability Strategy Office for Disability Issues. (n.d.). Retrieved June 5, 2019, from https://www.odi.govt.nz/nz-disability-strategy/
- Oliver, M. (2013). The social model of disability: thirty years on. *Disability & Society*, *28*(7), 1024–1026. https://doi.org/10.1080/09687599.2013.818773
- Parker-Katz, M. B., Cushing, L. S., & Athamanah, L. (2018). Fostering Collaboration as Transition Specialists Through Community Conversations. *Journal of Disability Policy Studies*, 28(4), 244–254. https://doi.org/10.1177/1044207317739407
- Petrie, K., Devcich, J., & Fitzgerald, H. (2018). Working towards inclusive physical education in a primary school: 'some days I just don't get it right.' *Physical Education and Sport Pedagogy*, 23(4), 345–357. https://doi.org/10.1080/17408989.2018.1441391

Petroulakis, N. (2017). A lightweight framework for secure life logging in a smart environment.

- Protocol Convention on the Rights of Persons with Disabilities and Optional Protocol UNITED NATIONS. (n.d.). Retrieved from https://www.un.org/disabilities/documents/convention/convoptprot-e.pdf
- Ratliffe, K. T., Rao, K., Skouge, J. R., & Peter, J. (2012). Navigating the currents of change: technology, inclusion, and access for people with disabilities in the Pacific. *Information Technology for Development*, *18*(3), 209–225. https://doi.org/10.1080/02681102.2011.643207

- Seeking Research Participants? Graduate and Postdoctoral Studies Simon Fraser University. (n.d.). Retrieved June 10, 2019, from https://www.sfu.ca/deangradstudies/events/seeking-research-participants.html
- Seelman, K. D. (1993). Assistive Technology Policy: A Road to Independence for Individuals with Disabilities. *Journal of Social Issues*, 49(2), 115–136. https://doi.org/10.1111/j.1540-4560.1993.tb00923.x
- Selvaraj, J. (2015). Inclusive education in New Zealand: policies, politics and contradictions. International Journal of Inclusive Education, 19(1), 86–101. https://doi.org/10.1080/13603116.2014.907584
- Shakespeare, T. (2013). Disability Rights and Wrongs Revisited. Retrieved from https://ebookcentral-proquestcom.ezproxy.otago.ac.nz/lib/otago/reader.action?docID=1524194&ppg=20
- Simon Sinek Education Rethink. (n.d.). Retrieved June 12, 2019, from https://www.facebook.com/watch/?v=391476431473486
- Strengths and weaknesses of Quantitative and Qualitative Research | WeeTech Solution Pvt Ltd. (n.d.). Retrieved June 28, 2019, from https://www.weetechsolution.com/blog/strengths-and-weaknesses-of-quantitative-andqualitative-research

Tintarev. (2016). Personal Storty Telling Using Natural Language.

- Trainor, A. A. (2018). Community Conversation as a Method of Gathering and Analyzing Qualitative Data. *Journal of Disability Policy Studies*, 29(1), 2–6. https://doi.org/10.1177/1044207317739403
- What I've Learned about Goals and Objectives Listen Technologies. (n.d.). Retrieved June 10, 2019, from https://www.listentech.com/what-ive-learned-about-goals-and-objectives/
- Wilson, P. (2011). *Positve perspective on cloud security*. *16*(3–4), 97–101. Retrieved from https://doi-org.ezproxy.otago.ac.nz/10.1016/j.istr.2011.08.002

Wise, P. H. (2010). Emerging Technologies and Their Impact on Disability. In *Source: The Future of Children* (Vol. 22). Retrieved from https://about.jstor.org/terms

World Health Organisation - What we do. (n.d.). Retrieved June 7, 2019, from https://www.who.int/about/what-we-do

Zuckerman, O., Gal-Oz, A., Tamir, N., & Kopelman-Rubin, D. (2015). *Initial Validation of an Assistive Technology to Enhance Executive Functioning Among Children with ADHD*. https://doi.org/10.1145/2771839.2771901

# 7 APPENDICES:

7.1 Information for Research Participants

# **Project title:**

Researching the concept of an individualised website for people with high and complex support needs.

# Dear Participant,

My name is Bridget Meyer and I am a Master of Professional practice student at the Otago Polytechnic. I work as a lead adviser for a national disability sport organisation that promotes opportunities for people living with physical impairments.

I am inviting you to be a participant in my Master of Professional practice.

# **General Introduction:**

The overall goal of my Master of Professional practice is to research and demonstrate the concept/idea of a website that encourages the sharing and storing of information for individuals with high and complex support needs at their discretion.

I will gather feedback from the five participant groups about the concept/idea.

3 individuals with high and complex support needs and/or their parents/caregivers.

- 3 primary school teachers.
- 3 teacher aides.
- 3 health professionals.
- 3 personal carers

This research will explore ways we can preserve and future proof the nature of our collective work for those who are unable to advocate for themselves. The research will be a catalyst in conversations on how to support the micro world of people with high and complex needs and the use of technology with appropriate information being housed in one place and accessed easily at the discretion of the individual or their advocate.

Through this process I will look for solutions on how the rich information that is used to create meaningful opportunities can be preserved for future reference and shared if and when appropriate.

My objective for the research is to analysis how intuitive and user-friendly, if at all, a website concept needs to be for people to populate it with information. I will demonstrate the concept of using a web based platform with various individuals and stakeholder organisations within the sector.

Through this research I will learn whether a person centred website with secure portals would be feasible.

#### Table 4: Ethics application information

This project will be used as a way to initiate conversations with the wider community on the

benefits of incorporating technology and interprofessional collaborative practice for people with high and complex support needs.

#### What is the aim of the project?

To research the concept of an online system that could be used to ensure information was stored and shared securely with the appropriate parties at the discretion of an individual with high and complex needs.

# How will potential participants be identified and accessed?

Individuals and their support networks will be invited to participate in the research through an independent organisation. Once the participant has expressed an interest they will contact me directly and I will begin the process of making a time to meet and carry out the interview questions. I will not approach anyone directly.

# Who will be the participants?

I will carry out fifteen interviews. These will be a combination of

Three individuals with high and complex support needs and/or their advocates.

Three primary school teachers.

Three teacher aides.

Three medical professionals.

Three personal carers.

# What will my participation involve?

Should you agree to take part in this research you will be asked to meet for a one hour interview at a location of your choosing.

The interview will be a discussion around the concept of using a website to store and share information that is tailored specifically for a person who has high and complex support needs.

With your permission I will record the session using an audio recorder.

# How will confidentiality and/or anonymity be protected? - see Otago Polytechnic Research Guidelines to determine whether you can promise these conditions.

All transcripts will be anonymous and no two participants will come into contact with each other. Names will be removed from transcripts and you will be given a number to code all correspondence. I will be the only person with knowledge of your transcript. Your transcript will be stored securely at Sport Otago's office.

Table 5: Ethics application information.

# What data or information will be collected and how will it be used?

The data will be collected through a series of interview questions (attached). There will be 7 questions in total and you can choose not to answer any question if you wish.

Results of this project may be published but the data included will not be linked to any specific participant.

You may request a copy of the results of the research and it will be available on request. This information will be available 10 months after the interviews.

# **Data Storage**

The data collected will be stored in such a way that only I as the researcher will have access to it. At the end of the project any personal information will be destroyed. The data will only be used for this project.

# Can participants change their minds and withdraw from the project?

You can decline to participate without any disadvantage to yourself of any kind. If you choose to participate, you may withdraw from the project up to 48 hours post interview, without giving reasons for your withdrawal. You can also withdraw any information that has already been supplied until the stage agreed on the consent form. You can also refuse to answer any particular question, and ask for the audio to be turned off at any stage.

# Additional support

Should you require additional support at any stage during this process this will be provided to

you at no cost to yourself. This may cover both sign language and language interpreters as well as counselling support if required.

You are also welcome to have whanau support accompany you.

# What if you have any questions?

If you have any questions about the project, either now or in the future, please feel free to contact

Bridget Meyer <u>meyerbt1@student.op.ac.nz</u> 027 6977177

Table 6: Ethics application information.

Dr Jo Kirkwood
Jo.kirkwood@op.ac.nz
021 1310 686
Any additional information given or conditions agreed to will be noted on the consent form which
will be a separate form with the title of the research on it.
I know that:
T KNOW that:
My participation in the project is entirely voluntary.
I am free to withdraw at any time without any disadvantage.
The data (including audio) will be retained in secure storage for five years after which time it will
be destroyed.
The participation in this project is entirely voluntary, there will be no remuneration or compensation provided.
The information gathered will be coded and anonymised replacing names with four digit numbers. The data collected will only be used in this research.
I agree to take part in this project under the conditions set out in the Information Sheet.
signature of participant
date
signature of researcher
This project has been reviewed and approved by the Otago Polytechnic Research Ethics
Committee.
Send a copy of the whole file to your manager to read, and the electronic copy signed by your Manager/HoS needs to be sent out to the Ethics Committee Administrator
103

(ethicsadmin@op.ac.nz).

Table 7: Ethics application information.

# 7.2 Ethics template to monitor changes

ISSUE: Research purpose and merit.	ADDRESSED BY:	Refer to	FEEDBACK FROM ETHICS COMMITTEE:	ADDRESSED BY:	Refer to page:
		page:			
It is not clear what the research hopes to achieve.	This project will test the creation of a demo website for people with high and complex support needs and their parents and caregivers. The action research project will give an insight into whether this medium of communication i.e the website is of value for people who are unable to advocate for themselves. The website will be tested with feedback on whether	Pg.3/4	Thank you. The clearest explanation for the lay reader is in the information to research participants section (appendices). The intention is clearest in the overall goal statement. Suggest that this wording is used through the document to provided further clarity as	I have brought the information from the Information for Research Participants across to the Ethics application to create consistency of language. I've reiterated that I am researching the concept of a site.	Рд 3
	is it something to be pursued or not.		it currently "shifts around".		
It is not clear from the title.	The research question has been changed to 'Does the ability to access a portal of individualised information improve the life of a person with high and complex needs'.	Pg 3	<ul> <li>Would suggest further refinement to the research question as you have set up a pilot. There also appears to be two distinct parts to the project.</li> <li>Having a title that encompasses both aspects, e.g., "Developing an interprofessional website for people with high and complex needs": and separating the content into using two research questions <ul> <li>One relating to the kind of information that would be helpful</li> </ul> </li> </ul>	I have now changed the research question to "Researching the concept of an individualised website for people with high and complex support <u>needs</u> "could add - "and the type of information that could be stored".?? Not sure.	Pg 3

			<ul> <li>the other, relating to the accessibility and usefulness of this information to the research groups might further clarify this project and enhance its findings.</li> <li>This would better convey the research intention and is supported by an action research approach.</li> </ul>		
High and complex	What is a 'complex' disability?	Pg 3	Thank you.	All good	Pg 3
needs to be defined.	The New Zealand Complex Carers				
	organisation describes a complex				
	disability as one that requires				
	intensive support:				
	• because the person has				
	multiple disabilities, i.e				
	physically disabled and				
	has an intellectual				
	disability				
	• is disabled and has				
	serious ongoing medical				
	condition, i.e epilepsy				
	• is disabled and has				
	autism spectrum				
	disorder or behavior				

Table 8: Ethics revision proof of changes.

	that requires a high level				
	of support				
Define the need for	This research will give an insight	Pg 4	OK, thanks for this.	All good	Pg 4
his research.	into the feasibility of taking this concept of an individualised website further. The demo site will enabled all parties to test the concept and to consider whether such a website with individualised portals will enhance the support they are providing people who require it.				
s there a perceived ap?	Research has been carried out identifying the need for support services to work more collaboratively. The New Zealand Complex Carers organisation identified in a 2017 report drew attention to the isolation carers are working in and the challenges faced in bringing support services together. Through my own role I am also faced with the yearly duplication of questions as student's transition from one class to another.	Pg 4	OK, this is clear and should be stated as a rational for the research.	Yes, I have done this.	Pg 4
Vhy this type of esource?	The ease of storing information over a cumulative period which does not involve support staff	Pg 5	ОК	All good.	Pg 5
	wading through a paper based resource is attractive. The instant accessibility to records/ideas/photos as well as the across platform sharing of ideas makes a website the most appropriate platform.				
Is there literature on this topic to support the application?	I have found a number of articles on the importance of a collaborative approach but I haven't found a lot of how to achieve such as approach.	Pg 5	ОК	All good.	Pg 5
Is there a pilot version of a website?	Yes, a demo version has been built. It has not been tested as yet.	Pg 5	ОК	All good.	Pg 5
Or is the researcher asking what kind of website/information would these different groups find useful?	As the bones of a website has been built as a starting point I am interested to know what information would be useful for all groups.	Pg 5	Ok, this relates to the earlier point about having more than one research question.	I have changed the research question so it makes reference to 'researching the concept of a website'.	Pg 5
What is the level of access to computers or apps among these groups?	This will vary from household to household however I imagine most do have access to some form of device.	Pg 5	OK. You may need to take an iPad for the demo.	Yes, I will use my laptop.	Pg 5
Would there be legal or privacy issues to be considered in setting up such a sharing website?	As the individual and the parent/caregiver will be the owners of the information which is housed in the portal they will decide who they invite to view the	Pg 5	This needs further clarification, particularly when dealing with vulnerable people. This mechanism is unclear, potentially unsafe/subject to misuse and possibly	As this is only a conceptualised idea I am gathering research on I do not believe this to be a barrier at such an early stage.	Pg 5

	information. Within each portal you require permission to view the various vaults. An example of this is that the sports coach would not need to see the medical vault however the medical professional might be invited to see the sports/leisure/recreation vault. All of these settings are managed by the host.		breaches and individuals person or group's privacy. Generic information (.e.g. that which is available in the public domain) is OK, However, including information relating to an individual's medical or personal situation, or anything that could be liked to an individual by inference (particularly as this is a relatively small and local community) cannot be shared.	I have left an additional paragraph explaining permission settings.	
ISSUE: Recruitment					
The recruitment method is unclear and difficult to follow.	The recruitment of volunteers will be managed through CCS – Disability Action Otago.	Pg 6	The new information sheet is better but requires further editing and simplification.	I have made changes to the flyer. I'm unsure if this is the information sheet you mean.	Pg 6
How will recruitment be done to select people who meet the inclusion criteria?	Step One: Flyer - advertising through CCS – social media, newsletter and other relevant organisations. Step Two: Letter of endorsement from CCS and participant information letter. Step Three: Interview times made and carried out.	Pg 7	Ok	Ok	Pg 7
Please clarify the recruitment process as this impacts on the ethical aspects/process of interviewing.	The recruitment of volunteers is being supported by CCS. A flyer will be made in the first instance advertising for volunteers to participate. The breakdown of the	Pg 6/7	ОК	Ok	Pg 6/7

	groups will be listed on the flyer as				
	well as the ages where applicable.				
Issues of vulnerable participants has been addressed but aspects of this need to be incorporated into the participant information and consent form.	These have now been included.	Pg 10	See previous comments, Still not clear.	Have added a statement in the participant information and consent form on where to get support.	Pg 10
ISSUE:					
Cultural/Language consideration:					
How have these been considered, especially in reaching different cultural groups (if relevant)?	I have elaborated on these.	Pg 8	Partially addressed.	I had made reference to the refugee programme and have added a comment of the access to an interpreter.	Pg 8
Please expand/reconsider this section in your application.	Have expanded.	Pg 8		Have added in language interpreter, sign language interpreter and whānau support.	Pg 8
Please consider the specific needs of the included groups throughout the revised application.	Have considered and revised.	Pg 8	Partially addressed.	Have added additional support options including counselling.	Pg 8

# 7.3 Letter of support from CCS - Disability Action:

You don't have to thank me - I am delighted for us to be involved!

Mel Smith

General Manager - Southern

DDI (03)479 6882

Mobile 0272 309 932

On 7/06/2018, at 2:56 PM, Bridget Meyer <<u>bridget@halberg.co.nz</u>> wrote:

Wonderful. Do you know how happy that makes me? Thank you so much. Ill be back in touch once I hear from Glenys and Jo.

Bridget

**Bridget Meyer** 

Lead Adviser

Halberg Disability Sport Foundation

E <u>bridget@halberg.co.nz</u>

C 027 6977177

L 03 474 6347

S bridget.halberg

----- Original message ------

# 7.4 Keeping track of required information

Document Title:	Action:	Other:	Notes:
Ethics application	Completed	Attached	Updated
Ethics revision table	Completed	Attached	Updated
Emailing from CCS confirming support.	Completed	Attached	No change
Draft flyer for research participants.	Completed	Attached	Updated
Interview questions	Completed	Attached	Updated
Letter of endorsement from CCS.	Completed	Attached	No change
Information for research participants	Completed	Attached	Updated

Table 9: Table to keep track of information required of journal entries
# 7.5 Sample of steps taken to analyse transcripts

	Teacher - AF	Student - AK	Support Worker - AR	Teacher - Beth Downie	Parent - CM	Parent - DM	Support worker -	Med Prof - IC	Med prof - JB	Parent - JK	Student - KD	Teacher - NB	Parent/student - RB	Med Prof - SG	Support Worker - TM	Med Prof - TW
What are you initial impressions?	Meeting a needfix s something we don't ourrently haveflocks easy to userhot adding to load/doesn't require huge amounts of training/fix seesaw but require huge amounts of training/fix energier hocificat prepier hocificat nos/moments to show of it is hugef filts a gaphvouldn't increase workload/a	Good concept/helpful to have stored in one place instead of all over the place where things get lost.	Fantastio concept/trailed off with ideas.	shaped/all in one place/ doesn't look complicated/in one central place/no	Vell liove itruseful resourceful about is on controll don't think there is anything ese out self explanatoryluse on any medium/different devices.	Could see how it oould wrk in the education setting/ every year ve lose information/have to reshate information the the time/information doesn't get shared/ without the having to be the medium all the time.	I think it is externely valuable/record of the person' see their strengths and weaknessesthis is something definitely lacking for me as a TArinformation lan't passed onfrind out last minute/one stop shop.	Simple and logical/real sense of person/lwould easily understand/under stand lwould be invited at times and not at other times.		system/SENCO has left/four different places that have different ways of recording/makes	allows persons to easily access their collated data/streamline	The verg impressed/consi stently wise - going forward for the child is very beneficial/ streamlining it/ going to be really productive.	I think it looks quite good actually bits of information everywherer I look back on old diaries, spreadsheetstr acking activities.	I'm very impressed/a long time ago multidisplinary teams who did share teams who did share professionals and none of therm meet/fills a gap/person oontrols access to krillis gap of storing information and storing securely.	I think it quite good because it good because it gets those professionals taking to each other as required of portfolio for of portfolio for kilds / seesaw programme.	Lifke the concept of it because often people are left having to reiterate all their information lots of times so the overall concept is good/seems straight forward.
Do gou believe the concept of a website where information could be both stored and shared among professionals would be helpful for both the individual and their support teams? If ges whg? If not,whg not?	Yes/Vrap around service/big teams/hot a person to help negotiate through that many people/communi action out could be really successfulit own of the come owned by one of these single entities/aducoati	YestThere are often a lot of support people/they do not need to repeat information if a new support person comes along.	Yesitts vital if it is unable to be contacted we need to act in the moment/ we need to know whatfit's all in is head/all the little things you take for granted/what provides quality of life a central place to be able	Yesiin one central placeino brainer	Yes/It becomes the one piece of information or database that people use/it becomes the place people know they need to go to to find out information/com munication in school/it gives her a voice/comms book some	Yest being able to hve access to stuff about herself.	Yesiget ideas on how to implement strategiesithis worked really wellfif you can get one idea out of it its' one idea you didn't have before.	is lost in translation/hadn't got the full understandingfim portant to be able to capture/incrediab le amount that become apparent about a goung perosn as theyg're growing up, it's really important to	Yesffish hook - time/parents in control which is a good thing.	Yesi gain information/read up and gain information before theg start trying to work with whoever.	Yesteasy access/made lost in trnslation/not repeating information/acce ss to important information.	Yestoollaboratio n/ different expertise.	Yes/ sharing information/ education/things get lost/so much to learn/hard to remember everything/peopl e to know about his interests/not usre how effective communication is between health professionals.	Yes/person with disability has the controlifirst time for me a repository of information.	Yes/professional s change and leave/new people come inflook back on history/what happened/whats been tried/good tohave all in one place.	Yes/information is siloed/have to reinterate information/not having to duplicate.
Do gou believe there would be ang positive outcomes from having access to a website such as this?	ication in huge	Manage our own profiles/hot having to repeat stuff.	information. High tumover of stafflittle things are important when you can't manage the big thingsfife's in much richer/see them as a social beings/net social beings/net/see beings/net/s	are embedded/you'v e had five years of creating/don't have to articulate	people don't read Continuity among professionals/ev ergone can access ithot reliant on us articulating that all the time/capturing life span.	information the better/orging out for information.	Different ideas/seeing what works/time poorlyou don't get planning time	keep track of. Limited ideas about what is possible/transition infin the current system is left behind/be really specific about skills/carry forward/all possibilities and interests that might have become apparent at one time early on and matbe	Our system doesn't have a lot of good stuff	Personal approach/knowin 9 background/not repeating stuff.	Control over information/sens e of freedom/quality of liefsupport team can access informatin quickly.	Good having information stored.	Information gets lost/better understanding of a persons needs.	telling story over and over.	Easier for people to talk to each other/angthing that adds to oornmunication	Background information/iii/id known that/personal attributes/build rapport.

Participant transcipt notes Themes one Question one Question two Question three Question four Question five Question six Quest ... (+) : (+)

	Teacher - AF	Student-AK	Support Worker - AR	Teacher-Both Daunie	Parent-CM	Parent-DM	Support worker - D	Mod Prof - IC	Modprof-JB	Parent-JK	Student-KD	Teacher - NB	Parentfrtudent- RB	Mod Prof - SG	Support Worker - TM	Mod Prof - TW
	wornst adding to	concept/helpfulto	Pentartic conceptL.trailed affuithidear	A vary z uccaziful tanif overything ir jurt blu af paper fyear chape of all in man place of daran' tanin man part of daran' tanin ma control place fon brainer fon kind of flass on.	Wall Have interaction research the person that it's about it an constraint dan't think there is anything also aut there dynamic hard explanatory for an any any adjumt different devices.	Omildree hau it caudi uurkin the aducation thingt avery year ue lare information have to rankree information all the time forfarmation aducen't get charad ubbast me having tabe the me dium all the time.	definitely lacking	Simple and logicalizedizate of personal survid early understand dranders tand usual be invited as timer and nat at wher timer.	Tablack N/2 a fast starting ide-off-affrom stars which is held with the shift and parasethehoud is a sile- transfer of the shift and the transfer of the shift and the independent indepen	The conception really good and findmatin already provide/flast information/ane- provide/flast information/ane- left/flaw different placer that have different supprofi- bear into the series arises - heap earlest	The concept is well designed one allow person to early access their callasted detafits comline detafits comline certain parale to access certain parts of fits.	Pm vory improved Acomplete anthy use - quine fearured for the child is vory baseficially atteamlining (int quine to be really productive.	I think it laske quite good actually file of information overywhere it lask back on ald diaring, are a dhoesttra aking activities.	I'm vary impraces of a lang time aga multi dipolia ary taona uha dipolia ary taona uha dipolia ary different prefaceimale and anna 6 them mostfills a gaptares antrale secare to thill a gaptares different information and staring its securely.	I think it quite quit because it quit that a professional that this is to ach when ar required an needed to card of particular for kider records professions professions	Hiko the consept of the course of the course of the relation of the relation of the off the relation of the time ray the more consept to quark research relight forward.
De yene de Université e concepte est a sucherine alter e la formantine cent d'a la la la la terrat de la constante en professional de un la charget fore back the individual and hairs regnert Neural? If yer shy? If watcaky wat?	increase unritined Arcimple usy of advacating faryourself. TardWrap around rerritering teamrinuts person to help negatists through that many person to help negatists through around the many person to help any of the many person to help the many person to help any of the many person to help any of the many the function of the the many person to help any of the many the function of the the many person to help any of the many the function of the the many person to help any of the function of the the function of the fu	Yor/Thoro are aften a last af ruppart pospie/thory da nat nood ta ropoat information if a neuropant for neuropant for neuropant	Yarfler vital if s is unable to be constant of un need to extin the mamment us need to know when the states when fitt at the little thing you take for grants of takes provide quality of field a central place to be oble to find information.	Fuhat pravidar quality of lifer a contra i place table able to find information.	Yorfft be camer the me piece all informations of database that people unofit becames the place people knowledge of the need to get to the find out information for scharfft gives her a varied camme bankrame people dan't seadiffdan't have a cameration	Yarfbeing abla ta koo accar tartuff abaut herrelf.	Yaariga (idaaran, haartu implemaar tarta tajarih uurika faaliy uurika faaliy uurika faaliy me idaa uuturik ito' me idaa uuturik dida't havo kafara.	Yerf information is last in translations/heads/t qut the full understanding/fing artens take ables a grower/incredials a growe	YoorWich hands - Vimarfyaron te in canval which is a quad thing.	Yarf quin informationales o d ya nd quin information bafero information bafero information baile information baile infor	Yartaary accurtenda available Fidear lartin translatinnfaat reparting information/accur x to important information.	Tarfeilleberetien different expertise	Yarizbaring informational advasation things qattartiza muchta isannihari ta konsu akawa hir interestrintzure hau effective cammunication it between health profeszionele.	Yorfporran uith dirailithy har the annraffirst and the farme a reparkary of information.	Yactyrafazinadr dhangand Levarhay paple cama inflank hay historytak happane di uhat haya all in ana place.	Yerfinfarmatian rilleafhava ta reinterate infarmatianfant havingta duplicate.
by gen believe there ameld be any mittice avecanes/rmm kering eccorr - underlie yeek or this?	Sharinqfaammuni catian in huqo taamu	Managa muraun profilorfact having taropostatuff.	High turnover of realfiftible bling are impartent uhen you can't industrie much richerizee them as a musical being riom tan being riom tand being riom tand canno ching?	Tranitionfidear are ambed de Fynu'we had five yearr af creatiopdan't hou to a striculate uhet they daar hou they da irtha flau an thraveh.	never demonstration of the second sec	Marc information the better/crying out for information.	Different ideertreing ubet uurkritime peertynu den't get planning time.	Limited idear about tabels purifield transition fin the current system is left behind its really specific about specific about still/read intercurr for use dial purificial block and specific about states and specific action of the specific action of th	Ovrzystem descrit News olst of good atviff	Perranal opprosch/finaming backgraum/fast repeatingstuff.	Oantralavor infarmatian/zontoj af free-fam/quality af life/ruppart team can accuz infarmatin quickly.	Gaad having information atored	Information, golf Iortho State understanding of a person needs	tellingstøry aver and aver.	Earlier for people to Valle to each an hard on y hing these adds to communication	Background information/if/ known that/porsonal attributor/build rapport.
In yuu baliava thara unuld ba eny aqutiva nutcumar frum having accarr u a uabrita such ar thir?	minimizing workload not adding to itfaxpoatationafar auting quidalineafparantz	Accorts informationfinfor mation gotting deleted.	Perranal intentfynu'll alwayr get peuple uhu have ne gativity touardr itfred tape.	Needr to be really user friendly fauick and eary / minute any thing is too hard, too complexbig	tao much offart no ano uror itfall traffic for cor people to uro it/X aunr itfall offart being pod it and it nat being urod.	Funding it f athors accessing information.	Giving pormizzian/nat having access ta device at schaol.	atherwise. Health information in degradingfreatrict ive about future/teachers and TAzee parzibilities	Safo from hackorrfhavo good profilor torhou hou to uro it/privacy.	Privacy izruartzian a waivor.	Parribly miruro of information/urong people quining accorr/getting people tartore in ane place could be an irreef staring	Safety of information/restri ctions were othered to.	Porzanal information romaine privato/privacy.	Canfidentail/priv acy and canfidential.	Internet quer duundree it ar anather jub tu du.	Security.

	P1	P4	P15	P2	P11	P13	P3	P7	P12	P5	P6	P10	P8	P9	P14	P16
	Harry	Ron	Hermione	Huey	Dueg	Luey	Rose	Lily	Violet	Fonzie	Chachi	Joanie	Dorothy	Rose	Blanch	Sophia
		Teacher - Beth				Parent/student -	Support Vorker -		Support worker -							
· · · · · · · · · · · · · · · · · · ·	Teacher - AF	Downie	Teacher - TM	Student - AK	Student - KD	RB	AB	Support worker - I	NB	Parent - CM	Parent - DM	Parent - JK	Med Prof - IC	Med prof - JB	Med Prof - SG	Med Prof - T
rson Centric / Empowerment									consistently vise		frecord of the		real sense of	information	fills a gap	
									going forward for	it's about is on	personi see their		person	should be held		1
									the child is very	control	strengths and			with the child and		1
									beneficial		weaknesses			parent/should be		1
														albe to selct who		1
	a simple way of													knows		1
	advocating for													what/positive		1
	yourself.													information		1
	yourser.													Information		<u> </u>
- Prins Red One sector star													rl would easily			
er Friendly / Comprehension																
	looks easy to									self			understandfunder			1
	usel doesn't									explanatory/use			stand I would be			1
	require huge									on any			invited at times			1
	amounts of	doesn't look			streamline					medium/different			and not at other			seems strai
	training	complicated			information?					devices.			times.			Forward.
	Its something we			/helpful to have it				this is something			every year we	rinformatin				
	don't currently			stored in one				definitely lacking			lose	alreadu				1
	have/ fills a gap			place instead of		bits of		for me as a			information/have	provided/lost				1
				all over the place		information		TAfinformation			to reshare	information/				1
				where things get		everwhere! I		isn't passed			information all	SENCO has				1
		everything is just		lost.		look back on old		on/find out last			the	left/four different			4/5 different	1
		bits of paper/pear		iosc.		diaries,		minute/one stop			timerinformation	places that have		what they do,	professionals	1
		shaped /no kind						shop.			doesn't get	different ways of				1
						spreadsheetstr		shop.				different ways of		how they do it	and none of	1
		of flow on.				acking activities.					shared/ without	recording		gets lost	them meet	
lenefits to individuals, profess, families	inot adding to				controls for										person controls	1
	workload/				allowing certain										access to it/fills	1
	moments to	all information in			people to access				streamlining it/			one system/			gap of storing	1
	show off	one place /in one			certain parts of				going to be really			makes life easier	1	one coherent	information and	1
	accessing ideas	central place	portfolio for kids		site.				productive.			heaps easier.		website	sharing it	
vironmental Context															a long time ago	1
											Could see how it				multidisplinary	1
											could work in the				teams who did	1
											education setting				share	
ecurity/Permission Settings																
ollaboration			I think it guite													
biaboration			good because it													
	save emailing 8		gets those													1
	people/notificati		professionals													1
																1
	ons#/accessibilit		talking to each													1
	y of it is huge		other as required													
															l	
	like seesaw but									/I don't think there						1
	extends to border	1	seesaw							is anything else						1
	people		programme.							out there						
		l	I	I	l	l	l	I		l	l			fish hook - time.	l	
nallenges		1	1	1	1	1		1								
hailenges																
allenges																

Table 10: Example of steps taken to analyse transcripts

# 7.6 Curriculum Vitae

Bridget Tracey Meyer

1032 Mount Cargill Road RD2 WAITATI 9085 03 482 1019 hm 027 69 77177 cell

Employment Pathway:

Qualifications:

Bachelor of Social Services (Distinction)

### **Current Masters Candidate**

**Topic** – Researching the concept of a storing and sharing website for individuals with high and complex support needs.

Employer:	Halberg Foundation
Date:	August 2005 – Present
Position Held:	Lead Adviser
,	Otago, Southland and Central Otago

### Key Performance Areas:

- Establish an extensive communication network with individuals in allied organisations both regionally and nationally within the social service sector.
- Effectively promote the Foundation's vision and mission in all presentations, newsletters and personal interactions
- Lead the Foundation's training courses and advisory services with knowledge, clarity and passion
- Advocate for young people with disabilities and ensure opportunities are created within the sport, recreation and leisure sector in an inclusive manner.
- Promote the work of the Halberg Disability Sport Foundation.
- Research, consult and analyse local community needs.
- Identify projects that are sustainable and coordinate the implementation of them.

Employer:	Adventure Development and Counselling.
Date:	January 2004 – August 2005.
Position Held:	Operations/Administration Manager.

### Key Performance Areas:

- To provide support and information on all programmes administered by Adventure Development and Counselling.
- Programming and planning of outdoor residential course.
- Market and promote this service to the wider community.
- Work in partnership with other at risk youth service providers.

Employer:	CCS Disability Action – Otago.
Date:	October 2000 – December 2004.
Position Held:	Recreation and Volunteer Coordinator.

### Key Performance Areas:

• Recruit, support, orientate and train volunteers to meet the responsibilities of their positions in supporting people with disabilities.

- Ensure volunteers are equipped with the necessary skills to undertake appointed tasks. Support volunteers, provide feedback on performance. Provide a job description that outlines the roles and responsibilities of the volunteer position. Coordinate volunteers to ensure outcomes are achieved.
- Develop and coordinate recreational/leisure activities.
- Network and liaise with relevant agencies.
- Monitor and address changing requirements of clients.
- Access and apply for grants through community funding organisations.
- Manage training and volunteer budget.

Employer:	CONNECT – Development Training and Corporate Events.
Date:	January 1999 – November 2003.
Position Held:	Self Employed/ Company Director.

Employer:	Otago Youth Wellness Centre.
Date:	January 1998 – August 1998.
Position Held:	Operations/Administration Manager.

Employer:	Roxburgh Children's Health Camp.
Date:	January 1992 – November 1997.
Position Held:	Fieldworker – Otago/ Southland

Employer:	Roxburgh Children's Health Camp.
Date:	January 1991 – January 1992.

Position Held:	Residential Child Care Worker.
Employer:	Cobham Outward Bound School.
Date:	February 1989 – January 1990
Position Held:	Catering / general Gopher

## **Voluntary Positions:**

2012 – Present	Koru Care Otago
2012 – Present	National Disability Snowsport Foundation
2011 – Present	Chairperson of the Otago Family Network.
2007 – Present	Blueskin Bay Youth Group coordinator.
2002 - 2003	CCS – Recreation support person for the Boccia team.
1990 – 1997	CanTeen – Otago/southland Coordinator.
	CanTeen – National Executive.

Achievements/Highlights:

Wednesday Sport Programme (2005 – present)

• An interschool sports initiative working with 8 secondary schools throughout Dunedin – an introduction to inclusive sports practice

Spike Swim Squad (2005 – present)

• A demand for young people with disabilities to have swimming lessons saw these classes piloted and then begin in an inclusive environment alongside able-bodied swimmers. These classes are

now administered through the Swimsation swimming class programme and run three times a week.

#### Jolt Dance (2007 – present)

• This initiative was created after a dance 'taster' at the Wednesday sport programme. The opportunity for creative rather than structured dance classes sees this programme running two classes a week.

#### Mustangs soccer team (2006 – present)

• The Mustangs soccer team is a mixed ability soccer team which plays throughout the season in the Unipol (University/Polytech) social grade competition. Supported by Physical Education Students.

#### Wildcatz Basketball (2006 – present)

• The Wildcatz Basketball team is a mixed ability basketball team which plays in Otago Basketball social league competition. It is now organised by PE students.

#### Otago Adaptive Snowsports (2007 – present)

• Collaboration between University of Otago and Halberg Disability Sport Foundation to create a sustainable pathway for athletes with disabilities within the snow sport arena.

#### Adaptive Outdoor Education (2008 – present)

• Collaboration between University of Otago, Halberg Disability Sport Foundation and Wild Earth Adventures to provide an outdoor experience for young people with disabilities as well as a learning opportunity for the Physical Education students who are supporting them.

#### Adaptive Sport Mentoring (2008 – present)

• Collaborative partnership between Regional Sports organisations, University of Otago and the Halberg Sport Opportunity Programme to create a sustainable pathway in inclusive sports for young people with disabilities.

### **Equipment Pool**

• An equipment pool has been established to allow families/schools/community groups to borrow adaptive equipment on a short term loan basis. Purchase of two 'Trailriders' imported from Canada have also been secured.

#### USA – sabbatical

• Visited a number of disability sport and recreation providers including University of Arizona disability sport programme – Tucson, Arizona, SPLORE (Special Population Learning Outdoor Recreation) Salt Lake City, Utah, Wounded Warriors Programme, Salt Lake City, Utah.

#### 'Attitude' programme on Central Otago Rail Trail

• Coordinate logistics for the filming of 'Attitude TV" Central Otago Rail Trail programme.

#### http://www.youtube.com/watch?v=kGy58BFo15I&feature=relmfu

Conference presentations – ISAPA (2011)

## Everybody In (2011)

The opportunity to present the concept of Adaptive Sport Mentoring at the International Symposium Adaptive Physical Activity (ISAPA) in Paris, France 2011

Outward Bound / Halberg Disability Sport Foundation

• A collaborative partnership between the Cobham Outward Bound School and the Halberg Disability Sport Foundation has been established with a pilot course being held in June 2014 for 10 physically disabled young people from throughout New Zealand.

Mango Tree Centre – Tonga

• A personal initiative to work alongside the Mango Tree Centre to create a disability sports programme and to introduce inclusive practices into primary schools on the island of Tongatapu.

### Personal Interests:

Personal Statement:	I am a firm believer in the philosophy 'it takes a village to raise a child'. I believe if you are in a position to support and advocate for others it is an honour and privilege to do so.
Family:	Partner and 2 sons (19 yrs and 20 yrs).
Interests:	Tramping
	Snow Boarding
	Gardening
	Sailing
Personal:	Warm, friendly, confident and enjoy working with people in a wide range of spheres.
	Reliable and trustworthy.
	Team player.
	Adaptable and patient.
	Organised, creative, with a can-do attitude.
	Have empathy for others and can participate in and/or lead situations that require diplomacy.

# 7.7 Review of Learning – Course one

#### Introduction:

I wanted my introduction into my Review of Learning to be filled with profound reflections, newly created coined phases and penny dropping epiphanies, however I have so many thoughts and questions racing around in my head it is difficult to know where to begin.

One thing I do have a lot of clarity on since finishing my Bachelor of Social Services however is that I want more! I want to learn more and I want to be more! I have many questions around my personal competencies as a productive member of my family and community and so many more questions around my profession, my current work place and the social inclusion space.

To link what I have learnt in the past to what my future aspirations are, I believe I need to firstly understand and gain more evidence in why I think the way I do when it comes to social inclusion -

not an easy task! I need to gain some perspective as to whether there is a space for what I believe is the right way forward or am I completely dreaming. I need to unravel and settle the neurons that are currently firing off in all directions and begin by having some conversations with individuals who are the recipients of the decision makers on what they believe is in fact social inclusion.

#### Professional:

If I look back to where I was at the beginning of my professional practice (30 years ago) there has been expediential growth. From being a reactionary fire fighter in my early days in the social service arena to what I believe now is a more strategic practitioner with a much greater understanding of the landscape in which I operate.

I am currently in the position of Lead Adviser for the Halberg Disability Sport Foundation. A position in which I lead a cluster of advisers from throughout the country. I also lead the recreation strategy at a national level which includes managing various projects with national stakeholders. These include Outward Bound and the Department of Conservation.

In completing my BSS my interest and curiosity into the space of social inclusion has created a brain explosion where currently I have more questions than answers! I believe by exploring this theme further I will be able find a place where my own philosophies will be challenged and hopefully reconciled. Upon the completion of this 'adventure', ie my Masters, I am looking to align myself with an organisation that has their compass set in the right direction - if there is a 'right' direction.

#### Personal/Education:

My personal journey and development to date has been quite a ride. In beginning my studies in 2016 and finishing them mid-way through 2017 was a cathartic experience and one, upon reflection, I am so thankful I was able to take. To be able to step outside of myself and be gifted the time and head space to create a timeline of 'happenings' and write about these in detail was truly decadent. To be given the support and encouragement to learn about ones hotspot and triggers and more importantly to then be supplied (actually, we paid a lot of money) with the tools

to manage these, has I believe, set me up to take make some major changes in my professional life.

Gaining a formal qualification that recognises and brings balance to both hemispheres of my brain has created a shift in my thinking and possibly 'a don't mess with me' attitude. I sometimes worry about the strength of my convictions around the inclusion model and would like to explore this more through my Masters. I have had over many years the opportunity to observe many different forms of best practice and have attended and organised many forums and conferences both nationally and internationally that touch on this topic.

#### Community/Volunteer:

My volunteering and community work has always been focused around my work. I have never explored or thought about why this is so up until now. I think perhaps we tend to be attracted to people who believe in the same as us therefore are drawn to similar social movements - whatever they may be. My take on volunteering is that if you can support others who are in need then this is what you ought to do. For the most part the giver will get more out of the act of giving than the receiver (unless you are doing periodic detention!).

I have also managed a lot of volunteers over the years and this too can come with its challenges. I have no doubt there is research on managing volunteers, but I don't think it is rocket science in working out how to do this well. My theory is that we form relationships with others/groups it is because we want to receive something back. Whether that is the sense of contributing to cause, a letter of reference, credits towards a degree, somebody makes you laugh or you have a social conscience that needs a top up. If we can provide an incentive for others to engage then the volunteering opportunity is usually a success.

My current volunteering is through an organisation called The Mango Tree Centre which is based in Tonga. (I have a social conscience that needs a top up and I have access to a lot of sports equipment that I can't stand to see be put in the landfill).

I have stepped down from chairing a number of groups over the last two years as I need to focus on other things including my personal and professional development.

120

#### Skills, Knowledge and Attributes:

My biggest strength or skill is also my Achilles heel! I am extremely solution focused which works superbly well for those I am advocating for but can be challenging to manage. I am fully aware of this trait and make a very conscious effort to pick my battle wisely. My knowledge and networks in the sector are vast and due to my solution focused approach to my practice if I don't know the answer to something I will find someone who does. I believe in surrounding myself with a scaffolding of knowledge.

In leading my team I also have the skills and knowledge to encourage and support others in managing both internal and external conflict. I am very aware that being in a middle management position that I am often pulled in both directions which can create its own challenges. I have learnt and put in place a number of tools that I use daily to manage and dilute the hysteria that can sometimes arise within our organisation. I am a firm believer that someone else's urgency is not mine and often find myself swiping left!

The attribute that has stood me in good stead and may seem a little out of place within the social service sector is I am creative with a relatively high level of detailed ingenuity. I can do both big picture thinking as well as detailed thinking which helps see projects and plans through from inception to completion. I am an ordered and visual thinker and use creative methods to remember various pieces of information. The challenging side to this within my own organisation is that it is very much disorganised with often the left hand not knowing what the right is doing. We receive mixed messages on a daily basis. I think when you are working with people as opposed to products you are going to get a melting pot of mixed messaging – this is the nature of the beast and rather than fighting it the best way forward is to put your own systems, whatever they may be, in place.

#### Future intentions on the MPP:

My past learnings and observations over the last five years is that my own organisation as well as many others including the Ministry of Education have lost their way on what social inclusion means. The Ministry of Education (Ministry of Education - Inclusive Education, Kāhore he rā) states that 'Inclusive Education' is whereby young people are "present, participating, learning and belonging". However, as time moves on we are becoming fully aware that some schools are not engaged in this value which means that students are moving away from their closest schools to attend schools where these values are upheld. This is creating a 'magnet school' model. The same is happening with the sport and club scene. Disability specific sport clubs are growing in popularity and the numbers that were once participating within a mainstream environment are dwindling. Through my studies I would like to understand why this is happening and look to do something to reverse this occurrence if possible.

I would also like to gain a much clearer understanding of why the mighty dollar that is provided by our funders and sponsors is influencing where we as an organisation put our resources. We are returning to the model of what is easiest not necessarily what is right! We have moved away from working hard alongside well established community events where we looked to break down barriers and encourage inclusion for people of all abilities. Our main focus now I believe is in running segregated/exclusive 'Have A Go' carrot dangling events once a year for physically disabled young people, whereby we take a number of 'money shots' (photos of children who are impaired i.e amputation, wheelchair users, cerebral palsy etc) which are then used as propaganda in our marketing material to appease our sponsors and funders and in turn generate more income. I have learnt that there must be a budget line in everything we do. I understand that we need an income stream to do our work but I am very uncomfortable having the dollars for services so closely linked. An example of this is a workshop called No Exception Training that we use to deliver for free to schools and sports clubs we are now required to charge for this information. There are enough challenges for this particular sector without creating more barriers. This may be brutal but I we are currently using other people's misfortune in a very unpalatable way! I do not agree that this is the way we ought to be conducting ourselves. I feel a fraud as this is not what inclusion is in my mind or in the mind of many of the families I work alongside!

I think I waited a whole 6 days after completing my degree before I made an appointment to meet with my supervisor again. I wanted to learn more! I didn't want this process of learning to stop. I wanted my ideas and ideals challenged and I found myself craving robust discussions on my

122

professional philosophies on inclusive practice. Inclusive Practice as stated in article 30 of the United Nation Convention on the Rights of a Person with Disabilities of which New Zealand is a member is outlined below –

Participation in cultural life, recreation, leisure and sport:

1. States Parties recognize the right of persons with disabilities to take part on an equal basis with others in cultural life, and shall take all appropriate measures to ensure that persons with disabilities:

(c) To ensure that persons with disabilities have access to sporting, recreational and tourism venues;

(d) To ensure that children with disabilities have equal access with other children to participation in play, recreation and leisure and sporting activities, including those activities in the school system;

(e) To ensure that persons with disabilities have access to services from those involved

(United Nations Convention - Right of Person with Disability, Kāhore he rā)

I grew an extra arm each time I held my own in various forums when I heard myself drawing upon theories, referring to research and articulating the latest findings in my area of interest. I can't help but think that my organisation is perpetuating social exclusion especially in that fact that our vision statement is 'An Inclusive New Zealand' - (Halberg Disability Sport Foundation, Kāhore he rā)

My future aspirations are to find the touch stone for what I stand for and find a space that I can practice this on/within. This process will be a positive and heathy platform that will allow me to sound out my theories, come up with a potential framework for practice and generate a personal guidebook that will give me direction for the future of my work practice.



Figure 20: Mind Map of Potential Journey of MPP process

# 7.8 Learning Agreement – Course two Master of Professional Practice

# Learning Agreement:

This Learning Agreement is between a Master of Professional Practice learner – Bridget Meyer and the Otago Polytechnic. The purpose of the Learning Agreement is to help ensure projects are completed in line with student learning goals.

# Name of Student:

Bridget Meyer

Facilitator:

Dr Glenys Forsyth

Academic Mentor:

Dr Jo Kirkwood

Title of Project:

'Research the concept of a storing and sharing website for people with high and complex support needs'.

# Introduction:

# 1. Context and background:

I am currently employed as a Lead Adviser for the Halberg Disability Sport Foundation, a role I have been doing for the past 13 years. My core role for the Foundation is focused on breaking down barriers and creating opportunities to encourage people of all levels of ability to participate in sport and recreation in a meaningful way alongside their peers. The strengths I bring to this role are in being innovative as well as being persistent around ways in which we can turn what some people perceive as an obstacle into what I would describe as an opportunity.

The context to which I am basing my Masters on has been formed over my career working within the social service industry and more recently within the disability sector. I have become more and more aware of the number of service organisations and personal that pass in and out of an individual's life each day, month and year. The personnel and

organisations I am referring to are teachers, teacher aides, physiotherapist, occupational therapist, personal carers and for some individuals their friends and family.

As this particular project does not align with the core work of the Foundation I am doing it independently of my employer. However, to achieve the necessary outcomes and to ensure the project has support I will be working with CCS – Disability Action Otago with additional support from New Zealand Complex Carers Group.

The opportunity for me is to find a way of using technology through a website where we can capture and share information that is specific to individuals with high and complex needs.

The definition of high and complex needs has been described by the Complex Care Group NZ as:

- Multiple disabilities including sensory disabilities, physical disabilities, severe intellectual disability, or serious and ongoing medical conditions.
- Behaviours that may require a very high level of support.
- A lack of appropriate services or policies to meet their intensive support needs.
- Often unable to voice their needs in appropriate forums due to their severe intellectual and learning disabilities.

### https://www.complexcaregroup.org.nz/voice/definintion-document/.

My interest in people with high and complex needs stems from conversations I have had with both individuals and their families, schools and community support organisations, on how information is stored and shared as a child transitions through their school years and post school. For individuals who are unable to advocate for themselves this information can be vital as they work towards creating a meaningful life experiences.

The situation that I have seen within my role revolves around the sourcing of ideas, and the planning and ability to implement these ideas. The time pressures to achieve certain tasks are challenging and things are often done in haste with few resources, training or guidance. The access to previous ideas in which a task may have been achieved with great success may have been archived in somebody's journal or are housed in the memory of a parent, support person, previous teacher or teacher aide. Many times people will think of a way to modify something, it will work well and then they will do nothing with that information. I also witness this regularly among my colleagues as I visit schools to work with young people with various levels of ability and discover the teacher aide working in a separate space from the rest of the class, having been given no information on what the class was about to engage in, few or no ques on what has worked the previous year. Therefore are unable to plan, modify or adapt in any meaningful engagement with the group. The sharing of this information that could potentially bring inclusive opportunities for an individual is often not shared as support staff continue to work in isolation.

The United Nations Convention on the Rights of Persons with Disabilities in article 30 talks about the need "to ensure that children with disabilities have equal access with other children to participation in play, recreation and leisure and sporting activities, including those activities in the school system". <u>Article 30 UNCRPD</u>.5d.

I was attracted to this project because of the repetitive enquiries I have received over my time with the Foundation about how an inclusive or workable opportunity has been created the previous season or year but then this information isn't easily retrievable for others to use.

A recent example of this was a school athlete's day for a young girl who has complex support needs. Rachel\* is a young girl who gets a huge amount of pleasure from being included with her able bodied classmates at every opportunity she is given. Rachel is unable to communicate in the traditional sense although she can make her feelings known. Rachel also has limited mobility. I received a phone call from school to say that their athlete's day was coming up and could I join them to help support the teacher aide in ensuring that Rachel was included. In Rachel's first three years of school she had been taken to the school library while the rest of the school attended the athletics day which is held off site. I have received the same phone call for the past 2 years. It appears there are new teachers and teacher aides supporting her each time. Nowhere has it been recorded what modifications were made for Rachel to participate alongside her peers. I believe there is an opportunity here to make an impact within this space. Had Rachel's teacher had easy access to images and ideas from last year's athletics day she would have been able to go into the day with a higher level of confidence on how to deliver a meaningful experience, the correct equipment and perhaps even a goal of looking to beat last year's times/throws. The integrity of the day would have been aligned to all of the other students participating in the athletics day.

# The following modifications were made for Rachel:

Whenever Rachel was required to jump or throw something we got her to use a small beanbag the size of her palm. For the long jump her teacher aide moved her chair to the end of the long jump pit and Rachel threw the beanbag into the pit from which the measurement was taken. We used the same idea with the high jump, the teacher aid moved her chair up to the bar and Rachel threw the beanbag over the bar. For the hurdles we lay them down flat and Rachel was able to step over them with support. Rachel experienced great success and a huge amount of encouragement from her classmates. To bring about any form of social change that is long lasting and to work towards building an inclusive community I believe we need to be able to access information easily. This information can often make or break an opportunity for an individual. Meanwhile children are being left behind while their classmates go on school camps, have swimming lessons, excursions to the beach, education outside the classroom modules among other things.

Below is a visual of how the website could potentially work.

- The site is person centered/controlled with different vaults of information housing different areas of a person's life.
- The person would decide the names of each vault.
- Professionals are invited by the person to be a part of the person's network or waka.
- Professionals are given permission to view different areas of a person's life.
- The professional uploads reports, photos, videos, pdf for others to view if permission is given.
- Nothing is shared without permission.
- A physio for example might be invited to view the sport, recreation, leisure and hobbies vault as there will be some relevant information in here, however the sports coach would not be given permission to view the health vault.
- This website would allow for all information to be stored on the one site.



2. Master of Professional Practice – Overall Goal:

# Aspirational Professional Practice Statement:

My intention is "to leave a legacy as someone who advocated and raised awareness of the importance of individual empowerment for people with disabilities".

The overall goal of my MPP is to introduce and test the concept of a website that encourages people to store and share information that is easily accessible to people with high and complex needs. In addition I will investigate whether the concept of a storing and

sharing website is something that could be supported by the sector and gather feedback on this.

Through this process I will look for solutions on how the rich information that is used to create meaningful opportunities can be preserved for future reference and shared when appropriate.

My goal for the website is to analyse how innate and user-friendly it is to populate with information and administer. I will assess the concept of using a web based platform with various stakeholders within the sector. I am interested to know whether a person centred website would be feasible or whether the site would be required to sit with an organisation. The success of the website will be measured in how instinctive it is to engage with for both the individuals and their support team.

This project will be used to initiate conversations with the wider community on the importance of incorporating technology and encouraging interprofessional collaborative practice. I will research ways we can preserve and future proof the nature of professionals work for those who are unable to advocate for themselves. The project will be a catalyst towards supporting the micro world of people with high and complex needs by using technology with all relevant information being housed in one place and accessed easily at the discretion of the individual or their advocate.

The interprofessional collaboration model of practice within the education setting has been described by Golom and Schreck when "people from two or more professions learn about, from and with each other to enable effective collaboration and improve outcomes" 2017 pg 3. To be a catalyst towards social change I believe that sharing, storing and creating a legacy of meaningful information is important. For individuals who are unable to advocate for themselves the long term benefits of having future proofed ideas and resources through interprofessional collaboration will assist in the building of an inclusive society.

#### 3. Learning Outcomes

- 1) To test the concept of an information storing and sharing website for people with high and complex needs.
- 2) Gather feedback and analyse.
- 3) To reflect upon my practice.

# 4. Main audience for my project:

There will be three main recipients who will benefit from this project.

- 1) The individuals with high and complex support requirements.
- 2) The family.
- 3) The professionals working alongside teachers, teacher aids, physios, occupational therapist, sport coaches and personal caregivers.

# 5. Action Research:

In looking at a number of different research methods I believe this particular project incorporates the main tenets of an action research project. Action research is practitioner based, self-reflective and improvement focused.

Three key components to an action research project that have come to the fore are:

- See a problem that needs a solution.
- Work collaboratively with others to solve problem.
- Motivate to improve the situation.

My reading on research methods pointed most favourably towards action research. Action research involves the researcher and encourages self-reflection as well as delving more deeply into an individual's own practice. An example of the questions I may use are 'what leads me to do the things I do?', 'What shapes the way I practice?'

My plan going forward using the action research model will look like the following:



Figure 21: Action research model

### 1) IDENTIFY THE PROBLEM:

The challenge that I have identified in my years as a practioner is the loss of valuable information among an individual's support network. This problem has also been highlighted in a number of reports and surveys carried out by the New Zealand Complex Care Group.

2) PROPOSE A SOLUTION:

To build a website that allows information to be stored for future reference as well as shared easily among an individual's support network.

3) TRY OUT THE SOLUTION:

Test out the concept of the website with stakeholders.

# 4) GATHER DATA:

Gather qualitative data from the group and look for common themes.

5) ANALYSE DATA:

Analyse the data and make changes if necessary. Go back to 1.

6) REPORT:

Write up a report with findings.

As a reflective practioner I am looking for a solution that I can share with the sector that will guide and support us to enhance the lives of the people we work alongside.

# Project Method:

The method I will use to collect data will be in a qualitative form. Qualitative research will allow me to capture information through a narrative story telling process.

- 1) I will conduct fifteen interviews. These will be a combination of three individuals with high and complex support needs and/or their advocates. Three primary school teachers. Three teacher aides. Three medical professionals and three personal carers.
- 2) A ten minute instructional webinar will be shown to the interview participants. Feedback on the webinar will be recorded. There will be set questions.
- 3) Participants will invited to use the website. Feedback will be recorded. There will be set questions.
- 4) The interview will be recorded. Each interview will be 1 hour in length.
- 5) The interviews will be carried out at a convenient time for the interviewee and at a location of their choice.
- 6) An independent organisation (CCS Disability Action Otago) will recruit participants to be a part of the research.
- 7) The qualitative data will be transcribed and the findings and conclusion written up in a report and presented.

# Project Methodology:

Action research methodology will allow me to be both researcher and reflective practioner throughout the project. The action research methodology will also allow me to work in a co-operative partnership with the research participants as well as my colleagues from the Foundation to look for solutions to the identified problems.

As a reflective practioner my awareness of my own biases will be front of mind. An example of bias are in the questions being asked, interpretation of feedback, how you ask questions, consistency in all the interviews.

Jean McNiff describes the action research methodology as "an enquiry conducted by self into the self. You, a practioner, think about your own life work, and this involves you asking why you do the things you do, and why you are the way you are".

I will keep a reflective journal that will shadow the more tangible results of the action research. At each stage of the project I will record my insights on why I thought or behaved in a certain way in conjunction with the steps being taken to carry out the project.



Figure 22: Example of parallel process

# 6. Ethical process:

My ethics application is being prepared for a category A application.

# 7. Māori Consultation:

My research does not target Māori specifically although will likely include Māori in the gathering of data. During my research I am also very interested to speak with the Ministry of Education regarding their Specialist Teaching programme called Māwhai. My understanding is that Māwhai is the Māori word web and net. This particular programme is a self-directed learning programme for specialist teachers to share the knowledge with one another in an educational setting only. I would like to gain an understanding of the protocols in regards to using Māori language as metaphors to describe scenarios and certain environments. I will make contact with the Māori adviser at the Otago Polytechnic.

Contact has been made – see below.

### Tēnā koe Bridget,

Thank you for your Masters of Professional Practice Capable NZ research proposal. My name is Kelli Te Maihāroa and I will be overseeing your application through the Kaitohutohu Māori consultation process. This email acknowledges receipt of your research application within our office. We look forward to learning more about your research proposal and will be in touch with you soon.

Nāku noa Nā Kelli Te Maihāroa

Tūmuaki Rakahau Māori Research Otago Polytechnic <u>Kelli.temaiharoa@op.ac.nz</u> T: 021493355

# 8. Literature Summary:

Introduction:

The framework for this review had been divided into two parts. I had begun the summary review by looking to form an understanding of segregation, the pros and cons of inclusion and then inclusion from a school point of view. For the second part of the review I had

begun to narrow in on what will form the foundations of my research project. However, after a reconfiguration on the scope of my project I have narrowed it down to focus on just the key elements and therefore have removed all of part one.

# PART 1

- 1) Segregation and social groupings.
- 2) Pros and cons of inclusive education.
- 3) Inclusive education.

# PART 2

- 4) Understanding Interprofessional Collaboration
- 5) Advocating for individuals with high and complex needs.
- 6) Lived Experience and empowerment.
- 7) Using technology in an unspoken world.
- 8) Transition of knowledge/ Storing and sharing sites.
- 9) Reflective practice.



My summary literature review is in this format as I did the orginal review using the reference programme of Endnote and in doing course three I moved to Mendeley due to what was avaiable.

PART 2

# 9. Sustainable Practice:

My overarching goal or outcome for this project is to test a concept, to house a dossier of tailored information for an individual that they will be able to use over their life time. I will test the concept, listening to the sector and responding to their feedback. The sustainability of this project and my practice within it will be measured in people's response to the concept of the website and their use of it.

I am motivated in my practice to always be looking for sustainable solutions on how we can better serve the population we are privileged to be working alongside.

# 10. Reflection:

I have begun a journal to capture conversations, emails, readings, my own thoughts and anything else that is relevant to my Master in Professional Practice journey. The journal will be provided as evidence where and when required.

# 11. Reflective Critical Commentary:

Through my reflective critical commentary journey I am going to be able to describe in detail what I have learnt, how I have learnt, why it matters and how this will influence my practice in the future. I will be able to draw and reflect upon the challenges as they arise and explain the processes that I undertook to overcome these moments. I will be able to describe in detail where my own beliefs and bias have been challenged. My reflective critical commentary will be a method of self-enquiry.

Insight One:

As this is only the beginning of my journey I have only just begun to capture the insights that are occurring alongside the writing up of project II. Already there have been a number of significant moments, particularly as I unravel the many lines of thoughts and the angles I could take with my supervisor. An example of the type of reflection and insight that is currently unfolding is the lofty scope I was initially trying to encapsulate within this project. After much discussion and contemplative thought I was able to see the importance of keeping the focus narrower but going deeper in my line of enquiry and not moving too quickly towards what I want the end result to be. Upon reflection I also understand and believe now that in narrowing the focus will ensure there is clarity.

I will continue to capture my reflective commentary as I move through this process.

(I am keeping a hand written journal of my insights and will add in time).

Insight Two:

Changing Tack...

Insight Three:

Conversation with Shelley at the end of 2017....

Insight Four:

Conversation with Dr Jude MacArthur beginning of 2018 around including the website as part of my research....

Insight Five:

Conversation with Jo and feedback from reviewers -....

Insight Six:

Conversation with Shelley and making it a work based project.....

Insight Seven:

Frustration...Halberg not supportive...back to plan A.

Insight Eight:

Changing tack for the third time....

## 12. Main outputs from my study:

The main output from this project will be a Practitioner Thesis:

1) The researching of a website based platform concept that will allow information to be stored and shared when appropriate.

#### 13. Key Milestones:

June 2018
June 2018
June 2018
July 2018
February 2018
March 2018

# 14. Employer/Professional representative: